ECONSTOR Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Braun, Sebastian; Dwenger, Nadja

Working Paper The local environment shapes refugee integration: Evidence from post-war Germany

Hohenheim Discussion Papers in Business, Economics and Social Sciences, No. 10-2017

Provided in Cooperation with:

Faculty of Business, Economics and Social Sciences, University of Hohenheim

Suggested Citation: Braun, Sebastian; Dwenger, Nadja (2017) : The local environment shapes refugee integration: Evidence from post-war Germany, Hohenheim Discussion Papers in Business, Economics and Social Sciences, No. 10-2017, Universität Hohenheim, Fakultät Wirtschafts- und Sozialwissenschaften, Stuttgart,

https://nbn-resolving.de/urn:nbn:de:bsz:100-opus-13605

This Version is available at: https://hdl.handle.net/10419/161927

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet. or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU





HOHENHEIM DISCUSSION PAPERS IN BUSINESS, ECONOMICS AND SOCIAL SCIENCES

Research Area INEPA

DISCUSSION PAPER 10-2017

THE LOCAL ENVIRONMENT SHAPES REFUGEE INTEGRATION: EVIDENCE FROM POST-WAR GERMANY

Sebastian Till Braun

University of St. Andrews

Nadja Dwenger

University of Hohenheim

State: May 2017

www.wiso.uni-hohenheim.de

Discussion Paper 10-2017

The Local Environment Shapes Refugee Integration: Evidence from Post-war Germany

Sebastian Till Braun, Nadja Dwenger

Research Area "INEPA – Inequality and Economic Policy Analysis"

Download this Discussion Paper from our homepage: https://wiso.uni-hohenheim.de/papers

ISSN 2364-2076 (Printausgabe) ISSN 2364-2084 (Internetausgabe)

Die Hohenheim Discussion Papers in Business, Economics and Social Sciences dienen der schnellen Verbreitung von Forschungsarbeiten der Fakultät Wirtschafts- und Sozialwissenschaften. Die Beiträge liegen in alleiniger Verantwortung der Autoren und stellen nicht notwendigerweise die Meinung der Fakultät Wirtschafts- und Sozialwissenschaften dar.

Hohenheim Discussion Papers in Business, Economics and Social Sciences are intended to make results of the Faculty of Business, Economics and Social Sciences research available to the public in order to encourage scientific discussion and suggestions for revisions. The authors are solely responsible for the contents which do not necessarily represent the opinion of the Faculty of Business, Economics and Social Sciences.

The Local Environment Shapes Refugee Integration: Evidence from Post-war Germany

Sebastian Till Braun^{*} Nadja Dwenger[†]

Abstract

This paper studies how the local environment in receiving counties affected the economic, social, and political integration of the eight million expellees who arrived in West Germany after World War II. We first document that integration outcomes differed dramatically across West German counties. We then show that more industrialized counties and counties with low expellee inflows were much more successful in integrating expellees than agrarian counties and counties with high inflows. Religious differences between native West Germans and expellees had no effect on labor market outcomes, but reduced inter-marriage rates and increased the local support for anti-expellee parties.

Keywords: Expellees; Forced migration; Immigration; Integration; Post-War Germany *JEL Classification:* J15; J61; N34; C36

Acknowledgement: Franziska Braunwart, Richard Franke, Philipp Jaschke, Laura Mockenhaupt and Daniele Pelosi provided excellent research assistance. We are grateful to Fabian Waldinger for providing us with data from the 1950 occupation census. We also thank David Krisztian Nagy, Guy Michaels, seminar participants at the University of St Andrews and participants of the workshop "Contextualizing the immigration debate: a historical perspective" at the University of Reading for their valuable comments. The research in this paper was funded by Deutsche Forschungsgemeinschaft (grant no. BR 4979/1-1, "Die volkswirtschaftlichen Effekte der Vertriebenen und ihre Integration in Westdeutschland, 1945-70"). Any remaining errors are our own.

^{*}University of St Andrews, UK, Kiel Institute for the World Economy, Germany, and RWI Research Network. Email: stb2@st-andrews.ac.uk.

 $[\]label{eq:constraint} ^{\dagger} University \ of \ Hohenheim, \ Germany, \ and \ CES if o. \ Email: \ nadja.dwenger@uni-hohenheim.de.$

1 Introduction

Does successful integration of refugees depend only on the innate characteristics of refugees, or is there also a role for the local environment of the host country? Does it matter whether refugees are re-settled in rural or urban areas of a host country, in small or large numbers, in culturally close or distant regions? All of these questions are central for designing refugee resettlement programs but have been largely overlooked in prior literature. This paper addresses the role of the local environment for integration in the context of one of the largest forced population movements in history, the mass displacement of ethnic Germans from Eastern Europe to West Germany after World War II. Eight million displaced persons arrived in West Germany between 1944 and 1950, most of them from the territories that Germany relinquished after the war. The integration of these expellees (*Heimatvertriebene*) was widely seen as the single most important challenge that the war-ridden country faced after 1945.

We empirically assess three hypotheses, formulated by contemporary social scientists, on the local determinants of expellee integration. First, we assess the argument that high population shares of expellees deteriorated integration outcomes. Second, we test whether rural and agrarian regions were less successful in integrating expellees than urban regions. Third, we evaluate whether religious differences between expellees and non-expellees had a negative impact on integration outcomes. Drawing on newly digitalized census and administrative data at the county level, we embrace a broad concept of integration: We use local labor market outcomes, inter-marriage rates between expellees and native West Germans, and electoral support for expellee and anti-expellee parties to measure the economic, social, and political integration of expellees.

Three features of our setting are important for the empirical analysis. First, economic, social, and political integration outcomes of expellees varied dramatically across West German counties. Looking at economic integration, for instance, expellees' labor-force-to-population ratio ranged from 31.6% to 59.0%. Second, West German counties were very heterogeneous in terms of their sectoral employment structure and predominant Christian confession before the expellee inflow as well as in the population share of expellees they received. That is,

the local environment that expellees encountered differed substantially. Third, our specific historical setting creates quasi-exogenous variation in the initial placement of expellees. The initial regional distribution of expellees was largely driven by the proximity to expellees' origin regions, and not by integration prospects (Connor 2007, Müller and Simon 1959, Nellner 1959): At the final stages of the war, ethnic Germans from East Europe fled from the approaching Red Army to nearby regions in West Germany. After the war, military governments in the West German occupation zones, overwhelmed by the size and pace of the inflow, were unable to distribute expellees according to their religious affiliation or local job prospects. Local German administration initially had no influence on the distribution of expellees. It is the quasi-exogenous variation in the initial placement of expellees that allows us to study the causal effect of expellee density, of the pre-war employment share in agriculture, and of religious differences between natives and expellees on integration outcomes.

While the occupying powers' military governments in West Germany did not distribute expellees according to their integration prospects, the local housing supply did influence the distribution. Given that much of the German housing stock lay in ruins after the war, the prime concern of the authorities was to provide expellees with a roof over their heads (Nellner 1959). Our empirical analysis thus controls for various indicators of war destruction to alleviate the concern that the local destruction level might have driven both the initial distribution of expellees and their subsequent integration outcomes. The initial, very unequal regional distribution of expellees persisted for several years after the war, as the occupying powers severely restricted relocations within Germany. In our empirical analysis, we use an instrumental variable (IV) strategy to address remaining concerns that expellees relocated endogenously within Germany after their initial placement. Our instrument isolates variation in regional expellee shares that is attributable to the initial placement of expellees and not to subsequent movements.

All of our analyses are based on unique historical census and administrative data which we were able to digitalize for the study. Our data set draws on statistics at the county level from the population and occupation censuses in 1939, 1946, and 1950, and the housing census in 1950. We further employ voting statistics for 1950, 1954, 1958 and 1962, sales tax statistics for 1935, marriage statistics for 1948 to 1952, and data on war destructions from the 1949 statistical yearbook of German municipalities and the county map (*Kreismappe*) of the Institut für Raumforschung.

Our main findings are as follows. First, the regional expellee share had strong negative effects on the economic, social, and political integration of expellees in 1950, i.e., five years after the war. A one standard deviation increase in the expellee share of a county decreases labor force participation of expellees by 0.4 standard deviations (or 5%), reduces inter-marriage rates by 0.3 standard deviations and increases the support for anti-expellee parties by 0.4 standard deviations (or 15%). This suggests a limited absorptive capacity of receiving counties. Higher expellee shares might intensify the tension between natives and expellees and make it easier for expellees to keep their own company.

Second, high shares of agricultural employment had an even stronger adverse effect on expellees' labor force participation: A one standard deviation increase in the share of individuals working in agriculture before the war reduces expellees labor force participation rate by 0.5 standard deviations (or 7.7%). Agricultural employment also worsened social and political integration outcomes but was less important for explaining regional differences in these variables. This suggests that resentments against expellees were higher in more agrarian regions, and that agrarian regions also had less capacity to absorb surplus population. The findings highlight potential costs of sending today's refugees to rural areas in order to avoid the formation of ghettos in the cities.

Third, differences in the religious confession between expellees and natives reduced intermarriage rates and increased the vote share of anti-expellee parties, but had no effect on expellees' labor market outcomes. This is consistent with the notion that shared values and traditions facilitate the social integration of refugees.

Fourth, political integration, the only dimension for which we have data over a longer time period, takes a considerable amount of time to complete. We find that the share of expellees and the religious distance between expellees and natives remain a strong predictor for the success of anti-expellee parties in 1954 and 1958. More than ten years after the arrival of expellees, a one-standard-deviation increase in the share of expellees still increases the vote share of the anti-expellee party by 2.0 percentage points.

Fifth, we show that the three factors we study-the regional population share of expellees, the pre-war employment share in agriculture, and religious differences between expellees and natives-explain a large part of the regional variation in integration outcomes. We find, for instance, that regional differences in the expellee share and in pre-war agricultural employment account for more than 60% of the variation in expellees' labor force participation. Overall, our results highlight that the local environment strongly shapes subsequent integration outcomes, and should thus be an important consideration when resettling forced migrants.

Related literature. Our paper complements a nascent literature that studies the distribution of (forced) migrants *across* countries, but generally abstracts from the effects on integration outcomes. Hatton (2015, 2016) argues that there is a strong case for a common asylum policy in the European Union (EU), but that such a policy can only reach the socially optimal number of admitted refugees if some form of financial burden-sharing exists. His arguments are based on a simple theoretical model of two symmetric countries, in which citizens value the admission of refugees to either country, but only face costs if refugees are admitted to their own country. Hosting refugees can then be viewed as an international public good that will be under-provided in the absence of cooperation.

Fernández-Huertas Moraga and Rapoport (2014) also start from the idea that hosting forced migrants creates costs for the host country and consider positive externalities for people who care about world poverty. They then show that tradeable immigration quotas can reveal country-specific costs of hosting migrants and thus each country's comparative advantage in hosting migrants. Since migrants typically have preferences over destination countries, and destination countries have preferences over migrants, Fernández-Huertas Moraga and Rapoport (2014) supplement the tradeable quota system with a matching mechanism that takes those preferences into account. Fernández-Huertas Moraga and Rapoport (2015a) discuss how such as a framework could work in the context of the Syrian refugee crisis and Fernández-Huertas Moraga and Rapoport (2015b) apply the framework to the EU Common Asylum Policy. They underline that EU countries trade quotas previously assigned to them through an allocation rule. Proposals for such allocation rules are widespread in the political debate. These rules typically calculate a country's "capacity" of hosting migrants based on economic criteria, such as population size, GDP per capita or the unemployment rate (see, for instance, Thielemann (2010) and European Commission (2015)). Such a rule, based on regional population and tax income, exists e.g. for the distribution of *today*'s refugees within Germany. Our empirical findings show how the regional distribution of forced migrants affects their subsequent integration outcomes, and can thus help to formulate evidence-based allocation rules.

Our result that expellee inflows increased the vote share for anti-expellee parties is consistent with recent evidence for Denmark. Dustmann et al. (2016) exploit quasi-random variation in the timing of refugee allocation, induced by a dispersal policy that randomly distributed refugees across Denmark. They find that outside urban municipalities, allocation of larger refugee shares between elections increases the vote share of anti-immigration and centre-right parties.

Damm (2009) exploits the same Danish dispersal policy to study the effect of ethnic enclaves, as measured by local ethnic concentration, on immigrants' labor market outcomes. The paper shows that seven years after their arrival, living in an ethnic enclave has a significantly positive effect on the earnings of refugees. Edin et al. (2003) also find a positive effect of living in an enclave on earnings of low-skilled refugees in Sweden, but not on earnings of high-skilled refugees. Our paper differs from the previous literature on ethnic enclaves in that we study the effect of the number of jointly resettled refugees on integration outcomes—and not the effect of the pre-existing local ethnic network. This distinction is likely to matter: Beaman (2012) shows for the US that the labor market outcomes of newly arrived refugees deteriorate with an increase in the number of recently resettled refugees of the same nationality.

Our paper also contributes to a small but growing literature on the economic effects of

displacement (reviewed in Ruiz and Vargas-Silva (2013)). Sarvimäki et al. (2009) study the long-term effects of the displacement of Finns from areas ceded to the Soviet Union after World War II. While they find a positive effect of displacement on the long-term income of male Finns who lived in rural areas before the displacement, the literature mainly documents negative economic effects of displacement. In post-war Bosnia and Herzegovina, employment rates are lower for displaced Bosnians than for Bosnians who stayed behind (Kondylis 2010), and displaced households in Northern Uganda experienced a significant decrease in consumption levels and asset values relative to comparable non-displaced households (Fiala 2015). Ibáñez and Vélez (2008) estimate that welfare losses caused by displacement within Colombia amount to 37% of the household's net present value of rural lifetime aggregate consumption. None of these papers studies how the displacement effect varies with characteristics of the initial resettlement location.

A few papers have exploited the quasi-experimental variation in our setting to study the effect of expellee inflows on structural change, native labor market outcomes, and regional population patterns in West Germany. Braun and Mahmoud (2014) document that large expellee inflows substantially reduced native employment in the short run. Braun and Weber (2016) consider a dynamic search and matching model to analyze how regional labor markets in West Germany adjusted to the inflow of workers over time. Braun and Kvasnicka (2014) show that expellee inflows fostered structural change away from low-productivity agriculture, but had a negative short-run effect on output per worker. Finally, Schumann (2014) uses a spatial regression discontinuity approach to show that the expellee inflow had a persistent effect on regional population patterns in the German state of Baden-Württemberg.

Regarding the economic integration of expellees, Bauer et al. (2013) compares the economic situation of expellees and native West Germans with identical pre-war observable characteristics. Their results show that in 1971, expellees and natives still performed strikingly different on the West German labor market (in line with earlier findings by Luettinger (1986)). In particular, expellees still earned significantly lower incomes than native West Germans and were over-represented among unqualified workers. Falck et al. (2012) show that the relative occupational position of expellees did not improve after the Federal Expellee Law (*Bundesver-triebenengesetz*) had been enacted in 1953, and hence conclude that the law did not achieve its aim of improving the labor market prospects of expellees. Whereas prior empirical literature provides important insights into the situation of expellees on the labor market, it neglects the importance of the local environment for integration, which is the focus of our study.

This paper is organized as follows. Section 2 provides background on the flight and expulsion of ethnic Germans from Eastern Europe. Section 3 explores regional variation in the integration outcomes of expellees, and outlines factors that can potentially explain these differences. Section 4 presents the empirical strategy and the data we use. Section 5 discusses our results, and Section 6 concludes.

2 The Flight and Expulsion of Germans from Eastern Europe

This section describes the flight and expulsion of ethnic Germans from Eastern Europe, the regional distribution of expellees in West Germany, and their socio-demographic characteristics relative to the native West German population. Henceforth, we will refer to those territories east of Germany's today's border that Germany lost after World War I or II as eastern territories. Figure 1 depicts Germany's territorial losses after the two world wars.

Flight and expulsion. Between 1944 and 1950, 12-14 million Germans were displaced from Eastern Europe. The displacement took place in three phases between 1944 and 1950 (for further details see Connor (2007), Douglas (2012) and Schulze (2011)).

The first phase of the displacement took place at the final stages of World War II and began when Soviet troops entered East Prussia in October 1944. The Soviet offensive on the East front prompted more than six million refugees from Germany's eastern territories to flee westwards (Oltmer 2010). Since the Nazis often delayed organized evacuations until it was too late, many people fled on their own. They either took the last train or ships out of the territories under attack or fled on foot. Refugees' initial destination in the West were largely



Figure 1: Germany's Territorial Losses 1919-45 and its Division in 1945

determined by the available escape routes (Müller and Simon 1959). Many East Prussians, for instance, rushed to the ports on the Baltic Sea and boarded ships that brought them to North Germany.

After Nazi Germany's surrender in May 1945, many refugees tried to return home. However, Polish and Soviet troops soon turned refugees away at the Oder/Neisse line (see Figure 1). At the same time, authorities in Poland–soon to be followed by those in Czechoslovakia– began expelling the remaining German population. These so-called 'wild' expulsions, which marked the second phase of the displacement, were not yet sanctioned by an international agreement, and continued until the end of 1945. Ethnic Germans were typically forced out of their homes on short notice and rounded up into holding camps. They were then either put on trains, or were marched to the border and driven into occupied Germany. While the

Base maps: MPIDR (2011).

number of ethnic Germans displaced during the wild expulsions remains unclear, existing estimates suggests that by the end of 1945, 800,000 to 1,000,000 people were displaced from Czechoslovakia alone (Douglas 2012).

The third phase of the displacement began in August 1945 when the Soviet Union, the United Kingdom and the United States concluded the Potsdam Agreement. The Agreement shifted the border between Germany and Poland westwards to the Oder-Neisse line. The eastern parts of Pomerania and Brandenburg, and most of East Prussia, were placed under Polish control. The rest of East Prussia went to the Soviet Union. German territories west of the Oder-Neisse line were divided into four occupation zones: a French zone in the southwest, a British zone in the northwest, an American zone in the south, and a Soviet zone in the east (see Figure 1). The three Western zones were later merged into the Federal Republic of Germany (henceforth: West Germany), the focus of our analysis. The Soviet zone became the German Democratic Republic (henceforth: East Germany).

The Potsdam Agreement of August 1945 also legalized the expulsions of Germans from Eastern Europe and stipulated 'that the transfer to [postwar] Germany of German populations, or elements thereof, remaining in Poland, Czechoslovakia, and Hungary, will have to be undertaken'. In November 1945, the Allied Control Council approved a timeline for the organized expulsion of the estimated 6.65 million Germans who were still living in Poland, Czechoslovakia, and Hungary at that time. The council also set quotas for the expellee intake of each occupation zone. Most of the organized expulsion transfers took place in 1946, but transfers continued on a smaller scale until 1950. Germans were either brought to holding camps or immediately put on often overloaded trains, which brought them to reception points in occupied Germany.

Regional Distribution. The flight and expulsion of ethnic Germans from East and Central Europe involved at least 12 million people. By September 1950, 7.876 million of them had settled in West Germany where they accounted for 16.5% of the population. The majority of them–around 4.423 million–had lived in the eastern territories that Germany ceded after



Figure 2: Population Share of Expellees in West German Counties, 9/1950

Notes: The figure shows the population share of expellees on 13 September 1950. The black line depicts the border of the three occupation zones. Source: Statistisches Bundesamt (1955). Basemap: MPIDR (2011).

World War II, namely Silesia (2.053 million), East Prussia (1.347 million), Pomerania (0.891 million) and Brandenburg (0.131 million). In addition, 1.912 million expellees came from the Sudentenland, the German-speaking part of Czechoslovakia which Nazi Germany annexed in September 1938. The remaining expellees had mostly lived in the eastern territories that Germany ceded after World War I, namely in Posen and West Prussia.

Importantly, expellees were distributed very unevenly across West Germany. Figure 2 depicts the county-level population share of expellees in September 1950. Three main facts stand out. First, the overall population share of expellees was much lower in the French occupation zone (6.6%) than in the American (18.7%) and British zone (17.2%). This was because the French initially refused to accept any newcomers into their zone. The French

had not been invited to the Potsdam conference. Therefore, they did not feel obliged to the commitment of the Potsdam agreement to secure an 'equitable distribution' of expellees across occupation zones.

Second, the population share of expellees was considerably higher in the eastern parts of the American and British occupation zones than in the western parts. This is particularly evident for the British zone where the expellee share was well above 30% in the north-east but as low as 5% in the far west. These enormous differences were mostly the result of the undirected flight of refugees at the final stages of the war. During this first phase of the displacement, refugees mostly sought shelter in those regions of West Germany that were closest to their former homelands and thus most accessible to them (Müller and Simon 1959). Many Germans from East Prussia, for instance, fled via the Baltic Sea to Schleswig-Holstein in the far north of Germany.

The 'wild expulsions' of the second phase only worsened these imbalances. Refugees were often just driven across the border into the eastern parts of occupied Germany. Germans from the Sudetenland, for instance, were often forced into neighbouring Bavaria. Even the organized transport of the third phase typically brought expellees to reception points in the east of each occupation zone.

Third, the population share of expellees was higher in rural areas than in cities. This was because many cities were in shambles after the war. Since housing was scarce, the military governments in the American and British occupation zones frequently restricted relocations into cities (Müller and Simon 1959). Instead, expellees were often housed in more rural areas where the housing stock had suffered less from bombing (Burchardi and Hassan 2013, Connor 2007). This rural-urban divide added to the regional imbalances, as the rural areas in the north- and southeast of Germany were already overburdened with refugees due to their geographical proximity to the eastern territories and the Sudetenland. It also explains why some of the smaller urban counties (*Stadtkreise*) in Figure 2 have low expellee shares despite being surrounded by larger rural counties (*Landkreise*) with very high shares.

The very unequal regional distribution of expellees remained largely unchanged in the first

few years after the war.¹ The occupying powers severely restricted the ability of Germans to change residence, and initially banned relocation altogether. After the ban was relaxed in 1947, moving still required permission from military authorities (permission was primarily granted for family reunification). It was not until the foundation of West Germany in May 1949 before the general freedom of movement was restored (Müller and Simon 1959, Ziemer 1973).

Socio-demographic characteristics. Expellees and natives were similar in several important respects. They both spoke German as their mother tongue and had both been educated in German schools. Moreover, the ceded eastern provinces, home to most expellees, had all been an integral part of the German Reich since the Reich was formed in 1871. Most expellees and natives had therefore lived in the same country for decades. Expellees were also not a selected sub-group of their home regions, as virtually all Germans living east of the Oder-Neisse line fled or were expelled.

As a result, socio-demographic characteristics of expellees and natives were similar. Table 1 shows that females outnumbered males both in the expellee and the non-expellee population, a legacy of the two world wars. Expellees were slightly younger, somewhat more likely to be single, and slightly better educated than the rest of the population. Overall, however, differences between expellees and natives were small, especially when compared to other migration episodes.

The mass arrival of expellees also had little impact on the denominational structure of West Germany as a whole. The shares of Catholics and Protestants were very similar in the expellees and non-expellees population (see again Table 1). However, the inflow of expellees had a significant effect on the denominational structure at a local level. As the expellees could not choose their initial destination based on the predominant Christian confession, and German authorities did not account for the religion of expellees when distributing them, many Catholic expellees ended up in predominately Protestant regions and vice versa (Connor 2007). In Bavaria, for instance, the number of exclusively Catholic or Protestant parishes fell from

¹The correlation coefficient between the county-level population share of expellees in 1946 and 1950 is 0.966.

	Expellees ^a	Rest of the population ^b
% females	52.9	53.2
Age structure		
% aged 0-17	29.7	27.7
% aged 18-24	11.3	10.1
% aged 25-44	30.0	27.9
% aged 45-64	21.8	24.6
% aged 65 and above	7.2	8.6
Marital status (aged 18 and above)		
% single	25.7	23.4
% married	60.4	64.0
% widowed or divorced	14.0	12.5
Education (born 1885-1927) ^c		
Years of schooling ^d	8.5	8.4
% vocational training	37.3	37.6
% university degree	3.5	2.9
Religious confession		
% Catholic	45.4	45.4
% Protestant	52.8	50.7
% Other	1.8	3.9

Table 1: Socio-demographic characteristics of expellees and non-expellees in West Germany, September 1950

Data sources: All data except for educational attainment are from the census of 13 September 1950, as published by Statistisches Bundesamt (1952). Figures on education are from our own calculations based on a 10% sample of the census of 27 May 1970 (FDZ 2008). Parts of the table are reproduced from Braun and Kvasnicka (2014).

Notes: ^a Expellees are defined as German nationals or ethnic Germans who on 1 September 1939 lived (i) in the former German territories east of the Oder-Neisse line, (ii) in Saarland or (iii) abroad, but only if their mother tongue was German. ^b The education statistics distinguish between expellees and native West Germans (excluding non-German foreigners). All other statistics distinguish between expellees and the rest of the population. ^c The education statistics are for those who were born between 1885 and 1927 (aged 23 to 65 in 1950). The overwhelming majority of these persons should have completed their education by 1950. ^d We only have data on the highest school degree. Years of schooling are inferred from the minimum years of schooling required to obtain a particular degree.

1,564 in 1939 to just nine in 1950 (Menges 1959).

Panel (a) of Figure 3 illustrates differences in the religious affiliation of expellees and nonexpellees at county level in September 1950. It depicts the Euclidean distance between the religious affiliations of expellees and non-expellees in county i:

$$Religious Distance_{i50} = \sqrt{\sum_{j} \left(share_{ij50}^{nat} - share_{ij50}^{exp}\right)^2},\tag{1}$$

where $share_{ij50}^{nat}$ ($share_{ij50}^{exp}$) is the share of natives (expellees) in county *i* who belong to confession *j*. We distinguish between Catholic, Protestant, and other religious affiliations.

Panel (a) of Figure 3 shows that the denominational structure of expellees and natives was relatively similar in the Protestant north of Germany, where mainly Protestant East Prussians arrived, and in the Catholic south-east, where many Catholic Sudeten Germans arrived. Differences were larger in western, middle, and south-eastern parts of the country. Many Catholic Sudeten Germans, for instance, were brought to settle in the mainly Protestant areas of North-Hesse and Franconia.

Panel (a) only depicts religious differences between the average expellee and native, but sheds no light on the overall effect of the expellee inflow on the denominational structure of a region. To capture how expellees changed a region's denominational structure, we consider the Euclidean distance between the actual denominational structure of a county, $share_{ij50}^{total}$, and the denominational structure of natives, $share_{ij50}^{nat}$:

$$ChangeReligion_{i50} = \sqrt{\sum_{j} \left(share_{ij50}^{total} - share_{ij50}^{nat} \right)^{2}} \\ = ExpelleeShare_{i50} \times \sqrt{\sum_{j} \left(share_{ij50}^{nat} - share_{ij50}^{exp} \right)^{2}}.$$
 (2)

The denominational structure of natives can be thought of as the hypothetical denominational structure of the region that would have prevailed without the inflow of expellees. Equation (2) illustrates that the overall change in the religious profile, $ChangeReligion_{i50}$, equals the denominational difference of expellees and non-expellees, $ReligiousDistance_{i50}$ (see above), times the regional population share of expellees, $ExpelleeShare_{i50}$.

Panel (b) of Figure 3 shows that the expellee inflow had the greatest effect on the denominational structure of Lower Saxony, Northern Hesse, and Franconia. In the North-Hessian county of Giessen, for instance, 94% of the native population but only 22% of the expellee population were Protestants. Since expellees made up a quarter of the total population, the overall share of Protestants in the county was 'only' 76%, down from 96.5% before the war. Changes in the religious profile were much more moderate in Schleswig-Holstein or Southern



Figure 3: Expellee Inflows and the Religious Profile of West German Counties, 9/1950

(a) Religious Distance

(b) Overall Change in Denominational Structure

Notes: The figure depicts the Euclidean distance between the denominational structure of a) expellees and non-expellees (Panel (a)) and b) the overall population and non-expellees (Panel (b)). See equations (1) and (2) and the corresponding description in the main text for more details. The black line depicts the border of the three occupation zones. The graphs divide the population into eight equally numerous subsets (octiles).

Sources: Own calculations based on Statistisches Bundesamt (1952). Basemap: MPIDR (2011).

Bavaria although these regions received very large inflows of expellees.

3 The Integration of Expellees in West Germany

The integration of eight million expellees into the West German economy and society posed a paramount challenge to the war-ridden country. This section presents descriptive evidence on the economic, social, and political integration of expellees in 1950, i.e., five years after the end of World War II. We show that the degree of integration varied greatly across West German counties. We then outline the factors that can potentially explain these differences, drawing on previous analyses of historians, sociologists, and contemporary observers.

Economic Integration. We consider the employment situation of expellees as our indicator for the *economic integration* of expellees, in line with contemporary observers (Connor 2007). We use the share of economically active persons in the expellee population (henceforth, labor force participation rate) as our main indicator and consider the share of employed persons in the population (henceforth, employment rate) as an alternative indicator.

Employment data come from the census of 17 September 1950. The census distinguished between economically active persons (*Erwerbspersonen*), independent economically inactive persons (*Selbständige Beruflose*), and dependent economically inactive persons (*Angehörige* ohne Beruf) (Statistisches Bundesamt 1955).² We calculate the labor force participation rate as the share of economically active persons in the total expellee population of a county.³ Importantly, there are many contemporary accounts that expellees, discouraged by dismal employment prospects, withdraw from the labor market and either retired early or returned to the fold (Pfeil 1958). The labor force participation rate captures this discouragement effect and can be precisely calculated for all West German counties.

The main drawback of the labor force participation rate is that it does not distinguish between economically active persons with and without employment. Although the census distinguished between the two groups,⁴ the German Statistical Office never published the

²Economically active persons are those who were in full-time employment at the time of the census or were looking for full-time employment. Part-time workers were not counted as economically active. Independent economically inactive persons were economically inactive but supported themselves through, in particular, retirement pensions or disability benefits. Dependent economically inactive persons were economically inactive and depended economically on another household member.

³ We cannot calculate the share of economically active expellees in the working-age population, as data on the expellee population by age is only available at the district level, but not at the more disaggregated county level. However, as a robustness check, we calculate a proxy for the county-level expellee population of working age by multiplying the district-level share of expellees aged 18 to 65 with the county-level expellee population. We then use this proxy to calculate the share of economically active persons in the expellee population aged 18 to 65. Section 5 shows that our conclusions are unchanged when using this variable as our measure for economic integration. This is to be expected as selection into specific regions was of no concern in our historical context, and regional differences in the age distribution of expellees were therefore relatively small.

⁴The census counted all persons as unemployed who usually carried out a full-time job but did not have employment at the time of the census. This includes persons not registered as unemployed at an employment

corresponding data at the county level and the original census records are, to the best of our knowledge, no longer available today. Fortunately, Pfeil (1958) drew on the original census records to calculate the share of economically active persons without employment (henceforth, unemployment rate), distinguishing also between expellees and non-expellees.

We use the data in Pfeil (1958) to calculate the employment rate, i.e., the share of employed persons in the population, as $(100 - \text{Unemployment rate}) \times \text{Labor force participation rate}$. Unfortunately, Pfeil only reports the unemployment rate in nine ranked categories, ranging from 0-4% to above 32%. We use midpoints of these categories to calculate the employment rate. Moreover, the unemployment rate is not available for the federal states of Südbaden and Württemberg-Hohenzollern, so that we can not calculate the employment rate for the 39 counties located in these two states. This is why we use the labor force participation rate rather than the employment rate as our main indicator of economic integration.

In West Germany as a whole, the labor force participation rate of expellees was 42.2% in September 1950. This is 4.2 percentage points lower than the participation rate of natives (46.4%). Differences between natives and expellees were even more pronounced with respect to the employment rate: 44.2% of the native population but only 35.9% of the expellee population were employed in September 1950.

Figure 4 illustrates that the aggregate numbers hide considerable regional variation in the labor market integration of expellees. The left panel shows that the labor force participation rate of expellees differs greatly across West German counties. It varies from 37.0% or less in regions in the lowest octile to 49.2% or more in the highest octile. There are clear regional clusters: Labor force participation is particularly low in the north, north-west, and south-east of West Germany and particularly high in the west and south-west of the country. These clusters are at times interrupted by the co-existence of small urban counties with higher participation rates and larger rural counties with lower participation rates.

The right panel, which depicts the employment rate of expellees in West German counties, reinforces these observations. The employment rate of expellees varies considerably between

office.



Figure 4: Labor Market Integration of Expellees in West German Counties, 9/1950

(a) Labor force participation rate

(b) Employment rate

Notes: The labor force participation rate is the share of economically active persons in the expellee population and the employment rate is the share of employed persons in the population. See the description in the main text for more details. The black line depicts the border of the three occupation zones. The employment rate is not available for counties in the federal states of Südbaden and Württemberg-Hohenzollern. The graphs divide the population into eight equally numerous subsets (octiles). *Sources*: Own calculations based on Statistisches Bundesamt (1955) and Pfeil (1958). *Basemap*: MPIDR (2011).

26.7% or less in regions in the lowest octile and 45.5% or more in the highest octile. Again, employment is particularly low in the north, north-west and south-east of West Germany and particularly high in the west and south-west of the country. The correlation between labor force participation and employment rates is 0.928.

Social Integration. Following contemporary sociologists (Müller 1950, Poepelt 1959), we use intermarriage rates between expellees and non-expellees as indicator for the social in-

tegration of expellees. Let a be the number of marriages between non-expellee men and non-expellee women in a region, b the number of marriages between non-expellee men and expellee women, c the number of marriages between expellee men and non-expellee women, and d the number of marriages between expellee men and expellee women (see Table 2). The indicator then compares the actual number of marriages between non-expellees and expellees, as given in Table 2, to the hypothetical number expected if the expellee status would not play any role for the choice of a spouse.

Table 2: Marriage behavior in a region

	Non-expellee women	Expellee women	Sum
Non-expellee men	a	b	a+b
Expellee men	С	d	c+d
Sum	a+c	b+d	a+b+c+d

Notes: Each entry gives the number of marriages in a cell.

Consider marriages between non-expellee men and expellee women. The actual number of marriages between non-expellee men and expellee women is b. The expected number is given by the probability of a randomly drawn men-women pair being a non-expellee man and an expellee woman, $(a + b)/(a + b + c + d) \times (b + d)/(a + b + c + d)$, times the total number of marriages in the region, a + b + c + d. The intermarriage rate between non-expellee men and expellee women is then calculated as:

$$\frac{100 \times b}{\frac{a+b}{a+b+c+d} \times \frac{b+d}{a+b+c+d}} \times (a+b+c+d) = \frac{100 \times b}{\frac{(a+b) \times (b+d)}{a+b+c+d}}.$$
(3)

Likewise, the intermarriage rate between expellee men and non-expellee women is:

$$\frac{100 \times c}{\frac{c+d}{a+b+c+d} \times \frac{a+c}{a+b+c+d}} \times (a+b+c+d) = \frac{100 \times c}{\frac{(c+d) \times (a+c)}{a+b+c+d}}.$$
(4)

The indicator varies between 0 (no marriages between expellees and non-expellees) and 100 (expellee status plays no role for the choice of a spouse). Higher values of intermarriage rates hence reflect better social integration. Importantly, the intermarriage rates calculated in equations (3) and (4) do not depend mechanically on the relative population size of expellees and non-expellees, as other commonly used indicators do (such as the share of marriages between two groups in the total number of married couples).

The average intermarriage rates across West German counties are 67.0 for expellee women and 71.9 for expellee men; expellees and non-expellees are significantly less likely to marry each other than what a random match suggests. Again, there is substantial regional heterogeneity. Drawing on data from Poepelt (1959), Figure 5 depicts county-level intermarriage rates in 1950–separately for expellee women and expellee men (data for Hesse and Schleswig-Holstein are only available at the federal state level). For female (male) expellees, the intermarriage rate varies from 52.7 (58.3) or less in regions in the lowest octile to 80.3 (85.0) or more for regions in the highest octile. Clear regional clusters arise: Intermarriage rates are particularly low in the south and in some parts of the north-west of West Germany and particularly high in the west.

Political Integration. The occupying powers harbored deep fears that the expellees could destabilize the young West German democracy–and thus placed strong emphasis on the political integration of expellees (Connor 2007). In fact, the Allies banned refugee organizations until the beginning of 1950, as they saw them as a potential source for the re-emergence of nationalism in Germany, and placed the responsibility of integrating expellees on the established parties. The established parties, in turn, were often reluctant to embrace expellee demands, as they feared losing the support of non-expellee voters. In fact, parties frequently campaigned on an outspoken anti-expellee stance.

The political integration of expellees can be studied from two perspectives, the electoral success of anti-expellee parties and that of expellee parties. Ideally, we would like to study a national election, in which both an anti-expellee and an expellee party competed for votes. However, expellee parties were still banned when West Germany's first national election was held in August 1949. Moreover, several parties only stood for election in a limited number of federal states, making it difficult to compare voting behavior across federal states.

Instead, we focus on the election for state parliament in Bavaria, one of the main refugee





Notes: The figure shows the intermarriage rates between expellee women and non-expellee men (left panel) and between expellee men and non-expellee women (right panel). See equations (3) and (4) and the corresponding description in the main text for more details on the calculation. The intermarriage rates are only available at the federal state level for the states of Hessen and Schleswig-Holstein. The graphs divide the population into eight equally numerous subsets (octiles). *Source*: Poepelt (1959). *Basemap*: MPIDR (2011).

states, in November 1950. The election offers three important advantages for our purpose. First, the expellee party *Bund der Heimatvertriebenen* (BHE) stood for election, forming an electoral pact with the right-wing nationalist party *Deutsche Gemeinschaft* (DG). The BHE primarily represented the interest of the expellees, demanding generous compensation for lost property and the recovery of the territories that Germany ceded after World War II. Second, with the *Bayernpartei* (BP), a fiercely anti-expellee party stood for election which articulated native Bavarian concerns of being swamped by foreign expellees (Connor 2007). In an infamous speech, Jakob Fischbacher, one of BP's founding members, called for the expellees to be thrown out of the country (Spiegel 1947). Third, the election date was very close to the date of the census, allowing us to relate regional vote shares to regional characteristics elicited in the census.

Figure 6 depicts the vote share of BP (left panel) and the combined vote shares of BP and BHE (right panel) in the Bavarian state election of 26 November 1950, as reported in Bayerisches Statistisches Landesamt (1951). State-wide, the BP received 17.9% of the vote, making it the third largest party in parliament (after the Social Democratic Party and the Christian Social Union). The BHE came fourth, receiving 12.3% of votes. In 15 out of 186 Bavarian counties, a majority of voters supported either the BP or the BHE.

The figure show that the vote shares for the two parties differed greatly across Bavaria. The BP was most successful in the south-east of Bavaria, reaching as much as 37.3% in Wasserburg am Inn. It was least successful in the north-west of the country where it frequently fell short of the 5 percent hurdle required to win seats in parliament. Adding the vote share of the BHE does not markedly change the picture. The combined share of the two parties were highest in the south-west and lowest in the north of Bavaria.

Explaining Geographic Differences in Integration. The great regional differences in the degree of economic, social, and political integration were not hidden to contemporary observers. Pfeil (1958), for instance, described the geographical location of the expellees as their 'destiny'. Likewise, there is no shortage of potential explanations for these stark differences in integration outcomes. What is missing, however, is a systematic empirical test of these explanations.

At least three not mutually exclusive hypotheses have been formulated. The first hypothesis states that high population shares of expellees were an impediment to local integration. The hypothesis holds that higher expellee shares intensified the competition on the labor market, slowing down the economic integration of expellees (Braun and Weber 2016, Pfeil 1958). Higher expellee shares might also have intensified the tension between natives and expellees



Figure 6: Vote Share of Special Interest Parties in Bavarian Federal State Election, 11/1950

Notes: The figure shows the vote share of BP (left panel) and the combined vote share of BP and BHE in the Bavarian state election of 26 November 1950. The graphs divide the population into eight equally numerous subsets (octiles).

Sources: Bayerisches Statistisches Landesamt (1951). Basemap: MPIDR (2011).

and made it easier for expellees to keep their own company (Connor 2007). This might have decreased inter-marriage rates. By slowing down economic integration, higher expellee shares might also have increased expellee support for the BHE. Moreover, the perceived threat of expellees to local traditions might have mobilized native voters to vote for anti-expellee parties.

The second hypothesis states that the integration of expellees was more difficult in rural and agrarian regions. The hypothesis holds that rural economies had little capacity to absorb surplus population, rendering the economic integration of expellees difficult (Connor 2007, Pfeil 1958). In fact, both Connor (2007) and Pfeil (1958) argue that the job prospects of expellees were determined less by the population share of expellees than by the 'absorption

capacity' of rural economies. It also has been argued that resentments against expellees were more pronounced in rural areas, and that relations between farmers and expellees were especially fraught with problems (Bayerisches Statistisches Landesamt 1950, Connor 2007, Schulze 2002). These tensions between natives and expellees in rural areas might be reflected in lower intermarriage rates and higher support for expellee and anti-expellee parties.

The third hypothesis states that religious differences between expellees and natives shaped integration outcomes. Qualitative regional studies indicate that expellees were more readily accepted in the predominantly Protestant state of Lower Saxony if they were Protestants themselves (Brelie-Lewien and Grebing 1997, Schulze 2002). Studies of Catholic Westphalia and Protestant Northern Hesse reach similar conclusions (Exner 1999, Spiegel-Schmidt 1959). Consequently, religious differences between expellees and natives might have slowed down the social integration of expellees, and might have increased the support for particularist parties. Religious differences might also have been an impediment to economic integration of expellees if they led to discrimination on the labor market.

Summing up the above, we have the following three testable hypotheses:

H1. Higher population shares of expellees deteriorated integration outcomes.

H2. More agrarian regions were less successful in integrating expellees than less agrarian regions.

H3. Religious differences between expellees and non-expellees worsened integration outcomes.

4 Empirical Strategy

We exploit regional variation across West German counties⁵ to test the three hypotheses. Our data come from various data sources that we have digitalized for our analysis. The sources

 $^{^{5}}$ While there are 556 of such counties in 1950, a few of them experienced changes in their administrative borders between 1939 and 1950. We account for these border changes by merging counties, so that county borders are comparable over time (see Appendix A for the details). This leaves us with 526 counties.

include the population and occupation censuses of 1939, 1946, and 1950,⁶ the housing census in 1950, administrative statistics on the Bavarian state election for 1950, 1954 and 1958, sales tax statistics for 1935, marriage statistics for 1948 to 1952, and data on war destructions from the 1949 statistical yearbook of German municipalities and the county map (*Kreismappe*) of the Institut für Raumforschung. Appendix B lists the data source for each variable.

4.1 OLS estimation

Let $Y_{i,50}$ be a particular indicator for the economic, social, or political integration of expellees in county *i* in 1950. Our basic regression specification is:

$$Y_{i50} = \alpha + \beta_1 ExpelleeShare_{i50} + \beta_2 A griculture_{i39} + \beta_3 ReligiousDistance_{i50} + X_{i39}\gamma + u_{i50},$$
(5)

where $ExpelleeShare_{i50}$ is the population share of expellees in county *i* in 1950, $Agriculture_{i39}$ is the agricultural employment share in 1939, $ReligiousDistance_{i50}$ is the religious distance between expellees and natives in 1950, X_{i39} is a vector of control variables for 1939 characteristics, and u_i is an error term. As counties vary widely in population size, we estimate population-weighted regressions (and provide unweighted regression results as a robustness check).

We consider three sets of integration indicators (see Section 3). First, we use the laborforce-to-population ratio of expellees as our main indicator for economic integration, and consider the employment-to-population ratio as an alternative indicator. Second, we use intermarriage rates between expellees and non-expellees, calculated separately for expellee men and women, as indicator for social integration. Third, we use the vote share for the anti-immigrant party *Bayernpartei* (BP) as our main indicator for political integration, and consider the sum of the vote share of the BP and the expellee party *Block der Heimatvertriebenen und Entrechteten* (BHE) as an alternative indicator.

⁶To the best of our knowledge, there exist no records of the underlying historical micro census data. Instead, we digitalized aggregated county-level data published mostly by the German Statistical Office.

The hypotheses H1, H2, and H3 imply that the three main explanatory variables of interest–expellee share, agricultural employment share, and religious distance–all have a negative effect on integration.

Population shares of expellees. Consider the expellee share first. Estimating equation (5) by ordinary least squares (OLS) will yield a consistent estimate of β_1 if $Cov(ExpelleeShare_{i50}, u_{i50}) = 0$. This covariance restriction implies that the expellee share must not be correlated with any unobserved factor that affects the economic, social, or political integration of expellees (depending on the outcome variable considered). In particular, the estimate of β_1 will be upward biased if expellees selected, based on unobservable characteristics, into West German regions where they saw higher chances of integration.

The problem of endogenous self-selection is arguably most severe with respect to economic integration, since the primary concern of expellees in the post-war period was economic deprivation (Connor 2007) rather than social or political exclusion. In fact, the inner-German migration of expellees in the 1950s were primarily movitivated by labor market prospects (Ambrosius 1996, Braun and Weber 2016). However, self-selection into thriving labor markets was arguably a minor problem until 1950 when we measure expellee shares. Importantly, the initial distribution of expellees was not driven by local labor market conditions (Braun and Mahmoud 2014, Nellner 1959). As described in Section 2, expellees first fled to regions close to their homelands, and were later transferred to their final destination region by the authorities. They could therefore not choose their destination based on local labor market conditions. Moreover, the occupying powers' military governments did not account for local job prospects when distributing expellees, and the local West German authorities, if functioning at all after the war, had initially no say in the distribution of expellees (Müller and Simon 1959). Once expellees had arrived in a region, they remained severely restricted to move elsewhere.

Overwhelmed by the size and pace of the inflow, the military governments' prime concern was to provide expellees with a roof over their head (Nellner 1959). Expellees were thus overrepresented in rural areas that were less devastated by the war (see again Section 2). If less destroyed areas offered better (worse) integration opportunities, this could potentially bias the effect of expellee density on integration outcomes upwards (downwards). Furthermore, moving restrictions were gradually phased out until 1949. Some expellees might therefore have moved endogenously to counties with better integration prospects by 1950.

We deal with these potential problems in two ways. First, we condition on various indicators of war destructions, and also on other local characteristics that may have influenced the integration prospects of expellees. Second, we use an instrumental variable (IV) strategy. We thereby isolate variation in regional expellee shares that is attributable to the initial placement of expellees and not to subsequent movements. We will discuss our control variables and the IV strategy in subsections 4.2 and 4.3, respectively.

Agrarian regions. Consider next the rurality of a region as measured by the agricultural employment share in 1939. For β_2 to have a causal interpretation, the pre-war agricultural employment share must not be correlated with the error term. We believe that this identifying assumption is likely to hold. In particular, reverse causality is of no concern since agricultural employment is measured in 1939 and thus before the arrival of expellees. However, agricultural employment might still correlate with unobserved determinants of expellee integration. We deal with this potential problem by controlling for regional characteristics that have been discussed as potential determinants of expellee integration (see subsection 4.2).

Religious differences. Focus finally on the religious distance between expellees and natives in 1950, as defined in equation (1). The main identifying assumption for a causal interpretation of β_3 is that religious distance must be uncorrelated with any unobserved factor that affects the economic, social, or political integration of expellees. Reverse causality should again be of little concern since the religious denomination of expellees and natives were pre-determined and changes of confession uncommon at the time. However, had expellees chosen their destination themselves, a high degree of religious distance might correlate with, potentially unobserved, regional characteristics conducive to integration. After all, Catholic expellees would probably only move to a Protestant region if this region would offer them exceptionally good integration prospects.

Although endogenous moving decisions should be of little concern in our context (as we have discussed before), we can not completely rule them out either. We again deal with this problem in two ways. First, we condition on variables that might have affected expellee integration. Second, we use an IV strategy to isolate variation in religious distance that is attributable to the initial distribution of expellees and not to subsequent movements. We next discuss the control variables and then our IV strategy.

4.2 Control Variables

Our vector of control variables consists of regional characteristics that might have affected expellee settlement pattern and influenced expellee integration. First and foremost, we use rubble at the end of the war per capita in 1939 as a measure of war destruction, following previous work by Brakman et al. (2004), Burchardi and Hassan (2013) and Braun and Kvasnicka (2014). War dislocation might have had an effect on both integration and-through the availability of housing-on expellee settlement patterns. Data on the amount of rubble, published in Deutscher Städtetag (1949), is only available for the 199 largest West German cities. We aggregate the data to the county level, implicitly assuming war destructions to be zero in smaller municipalities.

In a robustness check, we use the share of dwellings built until 1945 that were damaged in the war as an alternative measure. This measure, based on data from Statistisches Bundesamt (1956), has the advantage that it is available at the county level. However, it is not a direct measure of war destructions, as it relates only to residential housing that survived the war and could accommodate residents in 1950. In a second robustness check, we use a dichotomous variable, published by Institut für Raumfoschung (1955), that measures the loss in housing space in three categories ('no or minor losses', 'substantial losses', 'very substantial losses'). The dichotomous variable, which–given the lack of a comprehensive Germany-wide statistic–is also endorsed in Müller and Simon (1959), is based on various administrative sources at the national and federal state level.

Second, we control for the share of a county's population in 1939 that lived in cities with at least 10,000 inhabitants to account for pre-war differences in urbanisation, drawing on data published in Statistisches Reichsamt (1940). City dwellers might be more open to 'newcomers', as they had more contact with people from different backgrounds than inhabitants of rural areas (Connor 2007). At the same time, urban areas were more likely to be devastated in the Allied bombing campaign and thus received lower expellee inflows after the war.

Third, we include a dummy for regions located at the post-war inner German border (distance smaller than 75 kilometers). The inner German border might have impaired (economic) integration outcomes as regions at the inner-German border experienced a disproportionate loss in market access after World War II (Redding and Sturm 2008). At the same time, regions at the inner-German border also experienced high inflows of expellees because of their geographic proximity to the former eastern territories of the German Reich (see Section 2).

Fourth, we add a dummy for whether the majority of a region was Catholic in 1939, based on data published in Statistisches Reichsamt (1941). Religious affiliation might have influenced voting patterns in Bavaria and might also be more generally correlated with economic outcomes (Becker and Woessmann 2009, Weber 1904/05).

Finally, we use state-level fixed effects to control for unobserved factors common to all counties located in a state. State-level fixed effects also account for unobserved factors at the occupation-zone level (as each state is located in just one occupation zone).

4.3 IV Estimation

Our regression analysis conditions on covariates that might have affected the initial regional distribution of expellees, and also on other local characteristics that may have influenced the integration prospects of expellees. This distribution was very persistent in the first few years after the war, since the Allies severely restricted the freedom of movement until 1949 (see Section 2). However, some expellees might still have endogenously moved by 1950, leaving

behind their initial destination. If this re-location is based on unobserved characteristics, which in turn affect expellee integration, OLS estimates of β_1 and β_3 might be biased. In particular, one might expect β_1 to be upward biased if expellees relocated to regions with greater employment opportunities.

To deal with potential endogenous self-selection in the late 1940s, we use an IV strategy and isolate the variation in expellee shares and religious distance which is due to the initial placement of expellees only. In particular, we use the expellee share in October 1946, when severe restrictions on mobility were still in place, as an instrument for the expellee share in September 1950. The first stage regression for the expellee share in 1950 is:

$$ExpelleeShare_{i50} = \eta + \kappa_1 ExpelleeShare_{i46} + \kappa_2 A griculture_{i39} + \kappa_3 ReligiousDistance_{i50} + X_{i39}\kappa_4 + v_{i50},$$
(6)

where $ExpelleeShare_{i46}$ is the population share of expellees in county *i* in 1946 and X_{i39} is the same set of covariates as in equation (5). The key identifying assumption of the IV regression is $Cov(ExpelleeShare_{i46}, u_{i50}) = 0$. The assumption states that (i) there is no unobserved factor that drives both Y_{i50} and $ExpelleeShare_{i46}$, and that (ii) the expellee share in 1946 affects integration in 1950 only through its effect on the expellee share in 1950. In addition, we need the expellee share 1946 to be relevant for explaining the expellee share in 1950.

In a similar spirit, we also isolate the variation in religious distance that is due to the initial placement of expellees. Recall that religious distance is measured as:

$$ReligiousDistance_{i50} = \left[\left(share_{i50}^{cath,nat} - share_{i50}^{cath,exp} \right)^2 + \left(share_{i50}^{prot,nat} - share_{i50}^{prot,exp} \right)^2 + \left(share_{i50}^{other,nat} - share_{i50}^{other,exp} \right)^2 \right]^{0.5}.$$

$$(7)$$

Our instrument replaces the 1950 share of expellees belonging to a certain confession with the correspondent 1946 share. Unfortunately, we do not have regional data on the religious mark-up of expellees who lived in West Germany in 1946. Instead, we use data on the origin regions of expellees and the pre-war shares of the different confessions in these origin regions. The data allow us to distinguish seven origin regions (Silesia, East Brandenburg, Pomerania, East Prussia, CSSR (Sudetenland), Poland, Danzig). Let $ExpelleeShare_{i46}^s$ be the 1946 share of expellees from origin region s among all expellees in region i and let $share_{39}^{s,j}$ be the 1939 share of the population in origin region s belonging to confession $j = \{cath, prot, other\}$. We then approximate the predicted share of expellees in region i belonging to confession j in 1946 as:

$$share_{i46}^{j} = \sum_{s} ExpelleeShare_{i46}^{s} \times share_{i39}^{s,j}.$$
(8)

In principle, non-expellees might also have moved endogenously after moving restrictions were abolished. To address this potential problem, we replace the 1950 share of natives in region i belonging to confession j, $share_{i50}^{j,nat}$, by the corresponding 1939 share, $share_{i39}^{j,nat}$.

Our instrument is then given by

$$ReligiousDistance_{i46} = \left[\left(share_{i39}^{cath,nat} - share_{i46}^{cath,exp} \right)^2 + \left(share_{i39}^{prot,nat} - share_{i46}^{prot,exp} \right)^2 + \left(share_{i39}^{other,nat} - share_{i46}^{other,exp} \right)^2 \right]^{0.5}.$$

$$(9)$$

The first stage regression for the predicted religious distance is:

$$ReligiousDistance_{i50} = \delta + \lambda_1 ReligiousDistance_{i46} + \lambda_2 ExpelleeShare_{i50} + \lambda_3 Agriculture_{i39} + X_{i39}\lambda_4 + u_{i50},$$
(10)

The key identifying assumption of the IV regression is $Cov(ReligiousDistance_{i46}, u_{i50}) = 0.$

5 Empirical Evidence

The figures in Section 3 show substantial heterogeneity in integration outcomes across regions and reveal clear regional clusters. The following section aims at explaining this heterogeneity in order to understand what hampers and what promotes integration. In particular, we explore how the population share of expellees, the rurality of the receiving region, and the religious distance between expellees and natives determine integration outcomes. We consider three dimensions of integration: economic, political, and social integration. Causal evidence for each of these three dimensions is presented in the following subsections.

5.1 Economic Integration

We start with the determinants of expellees' economic integration, where economic integration is measured by success on the labor market. In a first set of regressions, we use the labor force participation rate of expellees in 1950 as the dependent variable. For a start, we focus on the size of regional expellee inflows. Column (1) of Table 3 presents estimates from an OLS model that includes the population share of expellees in 1950 as the variable of interest as well as our set of control variables (see Section 4.2). The correlation between the share of expellees arriving and the share of expellees in the labor force is negative and statistically significant (column (1)). In other words, the more expellees settled in a county, the lower the share of those who became economically integrated in the labor market. The estimated coefficient of -0.305 implies that a one standard deviation increase in the 1950 share of expellees (s.d. 0.089) reduces labor force participation of expellees by 0.44 standard deviations, or 6.0% relative to the mean labor force participation rate across all counties.

In Figure C, panel (a) in the Appendix we draw the (unconditional) linear regression line on the scatter plot between the population share of expellees in 1950 and the labor force participation rate. The figure illustrates that the relationship between the labor force participation rate of expellees and their population share is approximately linear and not driven by outliers. As this is true for all of our independent and dependent variables of interest (panels (b) to (i)), we stick to linear specifications in all following regressions.
Dependent variable:	labor force p	articipation ra	te 1950					employment	rate 1950
	OLS	OLS	OLS	OLS	OLS	IV	IV	IV	IV
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Expellee share 1950	-0.305***			-0.158***	-0.155***	-0.186***	-0.257***	-0.281***	-0.330***
	(0.068)			(0.040)	(0.056)	(0.037)	(0.059)	(0.054)	(0.072)
Agricultural employment share 1939		-0.230***		-0.191***	-0.149***	-0.184***	-0.141***	-0.259***	-0.217***
		(0.028)		(0.025)	(0.018)	(0.024)	(0.017)	(0.021)	(0.022)
Religious distance 1950			-0.002	0.000	-0.007	0.004	0.000	0.057***	0.015
			(0.019)	(0.015)	(0.010)	(0.017)	(0.011)	(0.018)	(0.016)
Population share living in cities with at least 10,000	0.047**	-0.031*	0.056**	-0.021	0.011	-0.018	0.008	-0.026***	-0.017
inhabitants 1939	(0.018)	(0.017)	(0.024)	(0.014)	(0.009)	(0.014)	(0.008)	(0.009)	(0.011)
Rubble per capita 1939	1.126	1.091**	2.257*	0.704	0.464	0.638	0.335	0.655	0.580*
	(0.732)	(0.550)	(1.246)	(0.477)	(0.372)	(0.459)	(0.312)	(0.441)	(0.329)
Distance to inner German border is smaller than 75 km (0/1)	0.022	-0.010	0.002	0.003	-0.001	0.005	0.001	-0.011	-0.005
	(0.014)	(0.007)	(0.013)	(0.008)	(0.007)	(0.008)	(0.007)	(0.010)	(0.010)
Majority is Catholic in 1939 (0/1)	-0.010	0.007	-0.002	0.001	-0.004	0.000	-0.006	0.000	-0.004
	(0.009)	(0.007)	(0.009)	(0.008)	(0.005)	(0.008)	(0.005)	(0.009)	(0.007)
R^2	0.573	0.662	0.458	0.687	0.777				
Weak identification test (Cragg-Donald Wald F statistic)	•	•	•	•		1027.51	759.96	1038.50	766.90
Shea's Partial R^2 : expellee share 1946				•	•	0.920	0.842	0.925	0.850
Shea's Partial R^2 : predicted religious distance						0.799	0.751	0.813	0.766
State dummies	no	no	no	no	yes	no	yes	no	yes
Number of observations	526	526	526	526	526	526	526	487	487

Table 3: Baseline results - forced migration and labor force participation of expellees

Notes: In columns (1) to (7), the dependent variable is the labor force participation rate of expellees in 1950. In columns (8) and (9), the dependent variable is the employment rate of expellees in 1950. The IV regressions in columns (6) to (9) use the expellee share in 1946 and the predicted population-weighted religious distance as instruments for the expellee share 1950 and the religious distance 1950, respectively. Columns (5), (7), and (9) include dummies for each of the nine West German states. Regressions are weighted with population in 1939. * * *, ** and * denote statistical significance at the 1%-, 5%- and 10%-level, respectively. Standard errors clustered at the labor market region level are in parentheses. The weak identification test refers to the Cragg-Donald F statistic; critical values from Stock and Yogo (2005) suggest the instruments to be strong.

Next, we turn to our second variable of interest: the agricultural employment share in 1939. The correlation between pre-war agricultural employment and the labor market integration of expellees is negative and statistically significant (point estimate: -0.230, column (2)). This is consistent with the idea that agrarian regions had little capacity to absorb expellees, as the amount of agricultural land was limited (Grosser 2006). In terms of magnitude, a one standard deviation increase in the agricultural employment share in 1939 lowers expellees' labor force participation rate by 0.86 standard deviations.

In column (3), we consider the correlation between the labor force participation rate in 1950 and the religious distance between expellees and natives in that year. The results show no correlation between religious dissimilarity and economic integration (point estimate: -0.002).

As discussed earlier, expellees were primarily settled in rural regions where more intact housing was available. In fact, expellee shares in 1950 and agricultural employment shares in 1939 are positively correlated. In a next step, we therefore include all three variables of interest in one regression model to separate the influence of expellee share and agricultural employment. The results in column (4) show point estimates that are somewhat smaller in absolute size. Still, we find that expellee inflow and a county's pre-war agricultural employment are economically and statistically significant determinants of economic integration. The coefficients of our control variables remain insignificant: Neither did counties at the inner-German border perform worse than other counties in terms of economic integration nor did rubble per capita affect expellees' labor force participation.

In column (5), we probe the robustness of our results and add fixed effects for the nine West German states to the set of control variables. These state dummies purge any unobserved factors at the state level which might simultaneously affect our explanatory variables of interest and the integration of expellees into the labor force.⁷ That is, we exclusively use the within-state variation to identify the effect of the different explanatory variables on economic integration. The coefficient of the expellee share is virtually unchanged at -0.155. That is, a one standard deviation increase in the 1950 share of expellees reduces labor force

⁷Note that adding state dummies has the downside of removing a lot of variation from our data set: regressing the 1950 share of expellees on state dummies gives an R^2 of 0.67.

participation of expellees by 0.23 standard deviations (or 3% relative to the mean). The coefficient of the agricultural employment share is very similar to before and clearly reveals worse economic integration prospects in more agrarian counties: A one-standard-deviation increase in the agricultural employment share reduces labor force participation of expellees by as much as -0.56 standard deviations. We still find no effect of religious distance on the economic integration of expellees in Germany.

As discussed in Section 2, expellees did not sort with a view to economic integration prospects and faced very tight moving restrictions. To alleviate concerns that some expellees might nevertheless have endogenously moved by 1950, we estimate IV regressions. There are two instruments that we use. First, we use the expellee share in 1946 as an instrument for the 1950 expellee share. Second, we use the predicted religious distance calculated with 1939 and 1946 values as instrument for the religious distance observed in 1950. Our IV strategy thus isolates the variation in expellee shares and religious distance that is due to the initial placement of expellees only (see Section 4 for a discussion of the instruments and the identifying assumptions).

Columns (6) and (7) in Table 3 contain the IV regression results for the labor force participation rate as dependent variable (without and with state-fixed effects, respectively). These are our preferred specifications. The lower part of the table presents summary results for the first-stage regressions. The Cragg-Donald Wald F statistic varies between 759 and 1028, suggesting that we do not have a weak instrument problem (for critical values see Stock and Yogo (2005)). Both of our instruments are relevant as shown by Shea's partial R^2 above 0.7. The detailed first stage regression results in the appendix reveal a strong economical and statistical relationship between the expellee share 1946 and the expellee share 1950 as well as between the predicted religious distance and the actual religious distance 1950 (see Table D1).

Dependent variable: labor force participation rate 1950	calculated o	ver overall pop	oulation				calculated over population of working age	
	unweighted		with alterna measure for	with alternative measure for damage I		with alternative measure for damage II		
	IV	IV	IV	IV	IV	IV	IV	IV
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
Expellee share 1950	-0.142*** (0.029)	-0.164*** (0.047)	-0.169*** (0.035)	-0.212*** (0.059)	-0.186*** (0.034)	-0.240*** (0.073)	-0.232*** (0.049)	-0.377*** (0.081)
Agricultural employment share 1939	-0.134*** (0.012)	-0.131*** (0.011)	-0.180*** (0.019)	-0.143*** (0.017)	-0.201*** (0.031)	-0.139*** (0.017)	-0.153*** (0.027)	-0.198*** (0.024)
Religious distance 1950	0.011 (0.010)	0.003 (0.009)	0.005 (0.014)	0.001 (0.011)	0.003 (0.016)	0.001 (0.011)	0.004 (0.022)	0.009 (0.015)
Population share living in cities with at least 10,000 inhabitants 1939	0.004 (0.007)	0.005 (0.007)	-0.024* (0.013)	0.003 (0.009)	-0.020 (0.019)	0.009 (0.009)	0.036*** (0.010)	0.018 (0.012)
Rubble per capita 1939	0.338 (0.402)	0.394 (0.331)					0.206 (0.480)	0.499 (0.440)
Distance to inner German border is smaller than 75 km (0/1) $$	0.001 (0.005)	0.003 (0.004)					0.014 (0.011)	0.008 (0.009)
Majority is Catholic in 1939 (0/1)	0.003 (0.004)	-0.008** (0.003)	0.001 (0.008)	-0.007 (0.005)	-0.001 (0.008)	-0.007 (0.005)	-0.003 (0.012)	-0.013* (0.007)
Loss in housing space, 3 categories [reference category: minor losses]								
substantial losses			-0.007 (0.005)	-0.004				
very substantial losses			0.025***	0.017**				
Damaged dwellings 1945			(,	()	0.003 (0.018)	0.014 (0.018)		
Weak identification test (Cragg-Donald Wald F statistic)	1103.24	711.82	1005.04	751.37	289.30	291.42	1027.51	759.96
Shea's Partial R^2 : expellee share 1946	0.890	0.754	0.928	0.812	0.923	0.800	0.920	0.842
Shea's Partial R^2 : predicted religious distance	0.814	0.778	0.795	0.749	0.795	0.749	0.799	0.751
State dummies Number of observations	no 526	yes 526	no 526	yes 526	no 526	yes 526	no 526	yes 526

Table 4: Robustness checks -	forced migration a	and labor force participation of	f expellees
------------------------------	--------------------	----------------------------------	-------------

Notes: The dependent variable in columns (1) to (6) is the labor force participation rate of expellees in 1950 calculated over the overall population. The dependent variable in columns (7) and (8) is the labor force participation rate of expellees in 1950 calculated over the population of working age (aged 18 to 65 years). All regressions are IV regressions using the expellee share in 1946 and the predicted population-weighted religious distance as instruments for the expellee share 1950 and the religious distance 1950, respectively. Columns (2), (4), (6), and (8) include dummies for each of the nine West German states. Regressions in columns (3) to (8) are weighted with population in 1939. ***, ** and * denote statistical significance at the 1%-, 5%- and 10%-level, respectively. Standard errors clustered at the labor market region level are in parentheses. The weak identification test refers to the Cragg-Donald F statistic; critical values from Stock and Yogo (2005) suggest the instruments to be strong.

The second stage results in columns (6) and (7) of Table 3 confirm that expellee inflows and the 1939 agricultural employment share had a highly significant negative impact on expellees' labor force participation rates-irrespective of whether we include state fixed effects or not. The coefficient estimates on the share of expellees are -0.186 (s.e. 0.037) and -0.257 (s.e. 0.059) without and with state fixed effects, respectively. The point estimate of -0.257 in the fully-fledged specification with state fixed effects suggests that an increase in the share of expellees settling in a county by one standard deviation reduces their labor force participation rate by 0.37 standard deviations or 5%. An increase in the agricultural employment share by one standard deviation worsens expellees' labor force participation rate by 0.53 standard deviations or 7.7% (point estimate with state fixed effects, column (7)).

To further probe the robustness of our results we use the employment rate as an alternative dependent variable (columns (8) and (9)). To wit, we focus on active persons with employment only (and disregard active persons without employment). Note that this information is available for a subset of 487 counties only so that we have to run the following regressions on a smaller subsample. Based on the regression without (with) state fixed effects we see a one standard deviation increase in expellee inflow to reduce employment by 0.30 (0.35) standard deviations, respectively. The corresponding results for the agricultural employment share are a reduction in employment by 0.71 (without state fixed effects) and 0.59 (with state fixed effects) standard deviations, respectively.

Table 4 present additional robustness checks to our preferred specifications (from Table 3, columns (6) and (7)). First, we run unweighted regressions in columns (1) and (2). And second, we use the loss in housing space in three categories (columns (3) and (4)) as well as the share of damaged dwellings (columns (5) and (6)) as alternative measures of war destruction. Reassuringly, in all of these robustness checks our point estimates hardly change. Finally, in columns (7) and (8), we use the labor force participation rate of expellees calculated over the population of working age, instead of over the population as a whole, as an alternative dependent variable. Since data on the expellee population by age is not available at the county level, we rely on data at the more aggregated district level (see Footnote 3 for details). We

again find that the expellee share and agricultural employment have a strong negative effect on labor force participation.

5.2 Social Integration

Next, we investigate the determinants of social integration. We measure social integration by the intermarriage rate between expellees and natives. Table 5 presents our core results on how expellee shares, pre-war agricultural employment of the receiving county, and religious distance affect intermarriage behavior. We begin by regressing the intermarriage rate separately on each of our variables of interest (conditional on our standard set of covariates). Column (1) shows an economically and statistically significant negative correlation between the expellee share in 1950 and the intermarriage rate. This is consistent with the view that higher expellee shares made it easier for expellees to keep to themselves and intensified animosity between natives and expellees (Connor 2007), both leading to lower intermarriage rates. Column (2) displays a significant negative correlation between the agricultural employment share in 1939 and intermarriage behavior, conditional on covariates. As hypothesised, more agrarian communities were thus less inclined to socially intermix (Bayerisches Statistisches Landesamt 1950, Connor 2007, Schulze 2002). We again do not find any relationship between religious distance and integration (column (3)).

In columns (4) and (5), we combine all explanatory variables in one OLS regression. The specifications in columns (4) and (5) differ in that state fixed effects are added in column (5), which thus only exploits within-state variation. Compared to the simple regressions in the first columns, the coefficients for both expellee and agricultural employment shares decrease somewhat but remain economically and statistically significant. An increase in the 1950 expellee share by one standard deviation lowers the intermarriage rate by 0.30 standard deviations or 6.9% (estimate of -0.366, column (5)). A one-standard-deviation increase in the agricultural employment share has about two thirds of this effect and decreases the intermarriage rate by 0.18 standard deviations (estimate of -0.079, column (5)). Note that the coefficient on the religious distance turns significant at the 5%-level once we only exploit within-state

variation by including state-fixed effects. The effect of religious distance is moderate: The point estimate suggests that a one-standard-deviation increase in religious distance reduces intermarriage rates by 0.11 standard deviations.

To deal with expellees who potentially moved endogenously based on unobserved factors, we again estimate IV regressions without and with state-fixed effects. Results are presented in columns (6) and (7), respectively. Differences between the OLS and IV estimation results are small and confidence intervals overlap. This suggests that endogenous self-selection of expellees into regions between 1946 and 1950 is a minor issue (in line with historical writings on this issue by, e.g., Müller and Simon (1959) and Ziemer (1973)).

So far, we have studied overall intermarriage rates. These overall rates, however, potentially mask important differences in social integration between male and female expellees. In columns (1) and (2) of Table 6, we therefore provide OLS and IV estimation results for the intermarriage rate between male expellees and female natives only. As we can see from the table, the social integration of male expellees is much more susceptible to the environment. In particular, an increase in the share of expellees arriving in a county markedly reduces the probability of male expellees to marry a female native: we find that an increase in the 1950 expellee share by one standard deviation lowers the intermarriage rate of male expellees by 0.46 standard deviations or 10.4% (OLS estimate of -0.553, column (1) of Table 6) compared to 0.30 standard deviations or 6.9% for all expellees (OLS estimate of -0.366, column (5) of Table 5). Also, religious dissimilarity appears to have been particularly detrimental for the social integration of male expellees.

	OLS	OLS	OLS	OLS	OLS	IV	IV
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
Expellee share 1950	-0.373***			-0.298***	-0.366***	-0.238**	-0.210*
	(0.072)			(0.093)	(0.115)	(0.100)	(0.122)
Agricultural employment share 1939		-0.164***		-0.086*	-0.079**	-0.102**	-0.092***
		(0.032)		(0.048)	(0.031)	(0.048)	(0.030)
Religious distance 1950			-0.073	-0.072	-0.057**	-0.069	-0.058**
			(0.056)	(0.052)	(0.024)	(0.057)	(0.026)
Population share living in cities with at least	0.084***	0.032	0.090***	0.048*	0.000	0.044*	0.006
10,000 inhabitants 1939	(0.017)	(0.020)	(0.019)	(0.026)	(0.017)	(0.026)	(0.017)
Rubble per capita 1939	0.890	1.349**	2.086***	0.617	0.675*	0.751	0.864**
	(0.617)	(0.677)	(0.733)	(0.615)	(0.356)	(0.630)	(0.399)
Distance to inner German border is smaller	0.055***	0.020	0.027	0.038*	0.045**	0.033	0.042**
than 75 km (0/1)	(0.019)	(0.018)	(0.017)	(0.020)	(0.018)	(0.020)	(0.017)
Majority is Catholic in 1939 (0/1)	-0.018	-0.004	-0.007	-0.008	-0.032***	-0.006	-0.029***
	(0.017)	(0.019)	(0.020)	(0.019)	(0.009)	(0.019)	(0.009)
R^2	0.349	0.328	0.308	0.373	0.609		
Weak identification test (Cragg-Donald Wald	F statistic)					714.167	218.586
Shea's Partial R^2 : expellee share 1946						0.916	0.847
Shea's Partial R^2 : predicted religious distance	2					0.760	0.722
State dummies	no	no	no	no	yes	no	yes
Number of observations	458	458	458	458	458	458	458

Table 5: Baseline results - forced migration and marriage behavior: all expellees

Notes: In all columns, the dependent variable is an index measuring expellee-native-marriages in 1950. The IV regressions in columns (6) and (7) use the expellee share in 1946 and the predicted population-weighted religious distance as instruments for the expellee share 1950 and the religious distance 1950, respectively. Columns (5) and (7) include dummies for each of the nine West German states. Regressions are weighted with population in 1939. ***, ** and * denote statistical significance at the 1%-, 5%- and 10%-level, respectively. Standard errors clustered at the labor market region level are in parentheses. The weak identification test refers to the Cragg-Donald F statistic; critical values from Stock and Yogo (2005) suggest the instruments to be strong.

Dependent variable: index for marriages between	male expelle	e and female native	expellees and natives						
			unweighted		with alternat measure for	tive damage I	with alternat measure for	ive damage II	
	OLS	IV	OLS	IV	OLS	IV	OLS	IV	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
Expellee share 1950	-0.553***	-0.456***	-0.486***	-0.360***	-0.469***	-0.345***	-0.400***	-0.262**	
	(0.116)	(0.125)	(0.080)	(0.096)	(0.121)	(0.123)	(0.110)	(0.118)	
Agricultural employment share 1939	-0.062**	-0.070**	-0.088***	-0.094***	-0.086**	-0.093***	-0.080**	-0.086***	
	(0.031)	(0.031)	(0.029)	(0.029)	(0.034)	(0.033)	(0.033)	(0.032)	
Religious distance 1950	-0.061***	-0.062**	-0.044**	-0.036*	-0.062**	-0.051*	-0.064***	-0.055**	
	(0.023)	(0.024)	(0.017)	(0.020)	(0.024)	(0.027)	(0.023)	(0.025)	
Population share living in cities with at least 10,000	-0.011	-0.007	0.004	0.009	-0.022	-0.018	-0.031	-0.027	
inhabitants 1939	(0.018)	(0.017)	(0.016)	(0.016)	(0.019)	(0.018)	(0.019)	(0.018)	
Rubble per capita 1939	0.670**	0.787**	0.586	0.728					
	(0.317)	(0.331)	(0.471)	(0.481)					
Distance to inner German border is smaller than 75	0.054***	0.052***	0.055***	0.055***					
km (0/1)	(0.017)	(0.016)	(0.011)	(0.010)					
Majority is Catholic in 1939 (0/1)	-0.030***	-0.029***	-0.045***	-0.042***	-0.034***	-0.034***	-0.038***	-0.038***	
	(0.009)	(0.009)	(0.010)	(0.010)	(0.010)	(0.010)	(0.010)	(0.010)	
Loss in housing space, 3 categories									
[reference category: minor losses]									
substantial losses					0.011	0.015			
					(0.011)	(0.011)			
very substantial losses					0.019	0.027**			
					(0.015)	(0.013)			
Damaged dwellings 1945							0.075**	0.094***	
							(0.031)	(0.030)	
R^2	0.614		0.495		0.591		0.598	•	
Weak identification test (Cragg-Donald Wald F statis	stic)	568.557		560.839		546.218		550.516	
Shea's Partial R^2 : expellee share 1946		0.847		0.745		0.819		0.812	
Shea's Partial R^2 : predicted religious distance		0.722		0.755		0.715		0.715	
State dummies	yes	yes	yes	yes	yes	yes	yes	yes	
Number of observations	458	458	458	458	458	458	458	458	

Table 6: Robustness checks - forced migration and marriage behavior

Notes: In all columns, the dependent variable is an index measuring expellee-native-marriages in 1950. The dependent variable in columns (1) and (2) is the marriage index for male expellee-female native-couples. The dependent variable in columns (3) to (8) is the index for marriages between expellees and natives. The IV regressions in columns (2), (4), (6), and (8) use the expellee share in 1946 and the predicted population-weighted religious distance as instruments for the expellee share 1950 and the religious distance 1950, respectively. All columns include dummies for each of the nine West German states. Regressions in columns (1) and (2) as well as columns (5) to (8) are weighted with population in 1939. ***, ** and * denote statistical significance at the 1%-, 5%- and 10%-level, respectively. Standard errors clustered at the labor market region level are in parentheses. The weak identification test refers to the Cragg-Donald F statistic; critical values from Stock and Yogo (2005) suggest the instruments to be strong.

As a robustness check, we additionally estimate unweighted regressions, taking the OLS and IV regression with covariates and state fixed effects on all expellees as our benchmark (columns (5) and (7) of Table 5). Columns (3) and (4) of Table 6 show the results. Point estimates are larger but statistically indistinguishable from the benchmark. This also holds true if we take the loss in housing space in three categories (columns (5) and (6)) or the share of damaged dwellings (columns (7) and (8)) as alternative measures of war destruction. Taken together, the results on intermarriage rates show that a stronger inflow of expellees, a more rural society and a larger religious distance hampered expellees and natives to socially intermix.

5.3 Political Integration

In a third set of results, we consider expellees' political integration. Our main indicator for (slow) political integration is the vote share of the anti-expellee party *Bayernpartei* (BP) in the Bavarian federal state election in 1950.⁸ We again start by regressing the vote share for BP separately on each of our three main variables (conditional on our standard covariates) and subsequently combine all variables into one OLS and IV regression, respectively. Results are reported in Table 7, columns (1) to (5). The vote share for the anti-expellee party increases with the population share of expellees, with pre-war agricultural employment of the county and with the religious distance between expellees and natives. The combined OLS (IV) regression suggests that a one standard deviation increase in the share of expellees adds as much as 2 percentage points (2.8 percentage points) to the vote share of the anti-expellee party BP. Given that the average vote share of BP was 18.6% at that time, this corresponds to an increase of 10.8% (15.0%). Correspondingly, a one standard deviation increase in the agricultural employment share (in religious distance) raises the vote share by about 4 percentage points (2 percentage points).

As an alternative dependent variable we consider the vote share for special interest parties, namely the sum of the vote share of the BP and of the expellee party *Block der Heimatver*-

⁸Note that the fact that we are exploiting a federal state election renders state fixed effects superfluous.

triebenen und Entrechteten (BHE). Columns (6) and (7) of the table show in an OLS and IV regression that the larger the share of expellees, the larger the agricultural employment share and the larger the religious distance between expellees and natives, the more are election outcomes driven by vested interests. In terms of magnitude, we find that a one-standarddeviation increase in the share of expellees leads to a 6.4 percentage points higher vote share for special interest parties and a one-standard-deviation increase in the agricultural employment share to an increase by 4.1 percentage points (calculations based on estimates from column (7)). A one-standard-deviation larger religious distance increases the vote share by 2.4 percentage points.

Our regressions on the vote share of BP are based on the vote share in the overall population, v_i^{pop} , that is, the number of votes for BP divided by the total number of votes. What we are eventually interested in, however, is the propensity of natives in a county *i* to vote for BP, v_i^{nat} , and how this share relates to the population share of expellees in that county, $ExpelleeShare_{i50}$. If we are ready to assume that expellees do not vote for BP and its antiexpellee election program, we can derive the share of natives who vote for BP by rewriting the population vote share for BP

$$v_i^{pop} = (1 - ExpelleeShare_{i50})v_i^{nat}$$
(11)

as

$$v_i^{nat} = \frac{v_i^{pop}}{(1 - ExpelleeShare_{i50})}.$$
(12)

This implies that the relationship between share of expellees and native votes for the antiexpellee party is even stronger than displayed in Table 7.9

⁹To see this consider the derivative of v_i^{nat} , $v_i^{nat'} = \frac{v_i^{pop'}}{(1-ExpelleeShare_{i50})} + \frac{v_i^{pop}}{(1-ExpelleeShare_{i50})^2}$ and compare it to the derivative of v_i^{pop} . As $v_i^{nat'} > v_i^{pop'} \forall ExpelleeShare_{i50} > 0$, we underestimate the effect by looking at the vote share in the overall population.

Dependent variable: vote share for	ent variable: vote share for Bayernpartei (anti-expellee party)					Bayernpartei + BHE (special interest parties)		
	OLS	OLS	OLS	OLS	IV	OLS	IV	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	
Expellee share 1950	0.461***			0.367***	0.509***	0.920***	1.159***	
	(0.122)			(0.116)	(0.129)	(0.130)	(0.141)	
Agricultural employment share 1939		0.210***		0.156***	0.139***	0.191***	0.162***	
		(0.036)		(0.036)	(0.036)	(0.044)	(0.044)	
Religious distance 1950			0.123***	0.105***	0.121***	0.082**	0.117**	
			(0.038)	(0.035)	(0.041)	(0.039)	(0.047)	
Population share living in cities with at least	-0.023	0.057**	-0.057***	0.055**	0.056***	0.030	0.033	
10,000 inhabitants 1939	(0.016)	(0.023)	(0.015)	(0.022)	(0.021)	(0.027)	(0.026)	
Rubble per capita 1939	-0.216	-2.122***	-1.001	0.342	1.073	1.369	2.666*	
	(0.780)	(0.465)	(0.754)	(1.111)	(1.314)	(1.031)	(1.381)	
Distance to inner German border is smaller than 75	-0.027**	-0.026**	-0.030**	-0.020*	-0.018	-0.005	-0.002	
km (0/1)	(0.011)	(0.012)	(0.012)	(0.011)	(0.012)	(0.012)	(0.013)	
Majority is Catholic in 1939 (0/1)	0.100***	0.084***	0.131***	0.129***	0.138***	0.128***	0.146***	
	(0.011)	(0.012)	(0.018)	(0.018)	(0.021)	(0.020)	(0.023)	
R^2	0.507	0.532	0.488	0.580		0.718		
Weak identification test (Cragg-Donald Wald F stati	stic)				250.45		250.45	
Shea's Partial R^2 : expellee share 1946					0.754		0.754	
Shea's Partial R^2 : predicted religious distance					0.775		0.775	
Number of observations	186	186	186	186	186	186	186	

Table 7: Baseline results - forced migration and political integration: state election in 1950

Notes: The IV regressions in columns (5) and (7) use the expellee share in 1946 and the predicted population-weighted religious distance as instrument for the expellee share 1950 and the religious distance 1950, respectively. Regressions are weighted with population in 1939. ***, ** and * denote statistical significance at the 1%-, 5%- and 10%-level, respectively. Robust standard errors are in parentheses. The weak identification test refers to the Cragg-Donald F statistic; critical values from Stock and Yogo (2005) suggest the instruments to be strong.

Dependent variable: vote share for Bayernpartei (national party) among non-expellees							
	OLS	OLS	OLS	OLS	IV		
	(1)	(2)	(3)	(4)	(5)		
Expellee share 1950	0.856***			0.730***	0.915***		
	(0.154)			(0.147)	(0.163)		
Agricultural employment share 1939		0.296***		0.204***	0.182***		
		(0.049)		(0.047)	(0.047)		
Religious distance 1950			0.145***	0.125***	0.149***		
			(0.051)	(0.046)	(0.052)		
Population share living in cities with at least	-0.033	0.063**	-0.096***	0.068**	0.070**		
10,000 inhabitants 1939	(0.020)	(0.032)	(0.022)	(0.028)	(0.027)		
Rubble per capita 1939	0.817	-2.716***	-1.391	1.429	2.410		
	(1.165)	(0.647)	(0.968)	(1.601)	(1.878)		
Distance to inner German border is smaller than 75	-0.038**	-0.039**	-0.046***	-0.029*	-0.027*		
km (0/1)	(0.015)	(0.016)	(0.017)	(0.015)	(0.015)		
Majority is Catholic in 1939 (0/1)	0.129***	0.101***	0.158***	0.163***	0.176***		
	(0.015)	(0.017)	(0.024)	(0.025)	(0.029)		
R^2	0.563	0.552	0.495	0.626			
Weak identification test (Cragg-Donald Wald F statis	stic)				250.45		
Shea's Partial R^2 : expellee share 1946					0.754		
Shea's Partial R^2 : predicted religious distance					0.775		
Number of observations	186	186	186	186	186		

Table 8: Additional results - forced migration and political integration: state election in 1950

Notes: The IV regression in column (5) uses the expellee share in 1946 and the predicted population-weighted religious distance as instruments for the expellee share 1950 and the religious distance 1950, respectively. Regressions are weighted with population in 1939. ***, ** and * denote statistical significance at the 1%-, 5%- and 10%-level, respectively. Standard errors clustered at the labor market region level are in parentheses. The weak identification test refers to the Cragg-Donald F statistic; critical values from Stock and Yogo (2005) suggest the instruments to be strong.

Given this insight, we use the approximated vote share of BP among natives, v_i^{nat} , as an alternative dependent variable. Results are shown in Table 8. Columns (1) to (3) again present the correlations with each of our main variables of interest (conditional on our standard covariates). Columns (4) and (5) display the results of OLS and IV regressions, respectively, that combine all of these variables. The estimated effect of the expellee share on the propensity to vote for BP for natives is about twice as large as is the propensity to vote for BP in the overall population which we estimated above. In the combined OLS (IV) regressions, we find that a one standard deviation increase in the share of expellees adds 4 percentage points (5 percentage points) to the vote share of the anti-immigrant party BP in the native population.

We complete our analysis on the political integration of expellees by studying the medium run vote shares for BP and for special interest parties. More precisely, we look at state election outcomes in 1954 and in 1958. That is, we analyze voting behavior more than ten years after the arrival of expellees. For each of the elections, we again consider the vote share for BP (the national party) and the vote share for special interest parties. OLS and IV results are reported in Table 9. We expect our variables of interest to lose explanatory power with time as expellees become more politically integrated. That is what we find for the expellee share and agricultural employment. While the coefficient on the expellee share in the BP regression is 0.509 in 1950 (Table 7, column (5)) it recedes to 0.365 in 1958 (Table 9, column (4)). Similarly, the coefficient on the agricultural employment share in 1958 shrinks to one third of its 1950 value and turns statistically insignificant. These findings reflect the advancing integration of expellees into West German society.

Dependent variable: vote share for	Bayernparte	i (national party	7)		Bayernpartei + BHE (special interest parties)				
	ir	n 1954	ir	n 1958	ir	n 1954	ir	n 1958	
	OLS	IV	OLS	IV	OLS	IV	OLS	IV	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	
Expellee share 1950	0.489***	0.562***	0.297***	0.365***	0.966***	1.066***	0.730***	0.821***	
	(0.100)	(0.108)	(0.086)	(0.092)	(0.119)	(0.121)	(0.095)	(0.103)	
Agricultural employment share 1939	0.024	0.017	0.047*	0.042	0.024	0.012	0.019	0.011	
	(0.032)	(0.032)	(0.028)	(0.027)	(0.038)	(0.037)	(0.032)	(0.031)	
Religious distance 1950	0.119***	0.122***	0.111***	0.104***	0.104***	0.116***	0.098***	0.094***	
	(0.034)	(0.037)	(0.031)	(0.032)	(0.037)	(0.042)	(0.033)	(0.036)	
Population share living in cities with at least	0.012	0.013	0.018	0.020	0.004	0.005	0.011	0.013	
10,000 inhabitants 1939	(0.019)	(0.018)	(0.015)	(0.014)	(0.022)	(0.021)	(0.018)	(0.017)	
Rubble per capita 1939	1.773	2.101*	1.671*	1.894**	2.395**	2.913**	1.987**	2.320**	
	(1.145)	(1.228)	(0.910)	(0.954)	(1.124)	(1.267)	(0.838)	(0.919)	
Distance to inner German border is smaller than 75	-0.053***	-0.052***	-0.043***	-0.043***	-0.044***	-0.043***	-0.039***	-0.038***	
km (0/1)	(0.010)	(0.010)	(0.008)	(0.008)	(0.012)	(0.012)	(0.009)	(0.009)	
Majority is Catholic in 1939 (0/1)	0.117***	0.120***	0.093***	0.092***	0.119***	0.126***	0.102***	0.103***	
	(0.018)	(0.020)	(0.015)	(0.015)	(0.020)	(0.022)	(0.016)	(0.017)	
\underline{R}^2	0.529		0.427		0.614				
Weak identification test (Cragg-Donald Wald F stati	stic)	250.45		250.45		250.45		250.45	
Shea's Partial R^2 : expellee share 1946		0.754		0.754		0.754		0.754	
Shea's Partial R^2 : predicted religious distance		0.775		0.775		0.775		0.775	
Number of observations	186	186	186	186	186	186	186	186	

Table 9: Medium-run results - forced migration and political integration: state elections in 1954 and 1958

Two points are remarkable though. First, comparing the 1954 to the 1950 regression results reveals that political integration takes a considerable amount of time: The coefficients on the expellee share and on religious distance in the 1954 regressions are very similar to the corresponding ones in the 1950 regressions (both OLS and IV). And second, the 1958 regressions show that even ten years after the arrival of expellees their political integration is far from being completed: the statistically significant coefficient on the share of expellees in column (4) suggests that a one-standard-deviation increase in the share of expellees still increases the vote share of the national party BP by 2.0 percentage points in 1958.

These patterns also become visible when we turn to the vote share for special interest parties as dependent variable (results reported in columns (5) to (8)). Our OLS and IV regressions provide evidence for deepening but sluggish political integration. In 1958 a onestandard-deviation increase in the share of expellees leads to 4.5 percentage points more votes for special interest parties (column (8)), compared to 6.4 percentage points eight years earlier. While the agricultural employment share is no longer a statistically significant determinant of the electoral support for BP and BHE, religious distance between expellees and natives remains a strong predictor for the success of these parties in 1958.

6 Conclusion

This paper contributes to the growing literature on the integration of forced migrants. In particular, we provide novel insights on the local factors that impede or facilitate the integration of forced migrants. Such insights are crucial for designing resettlement programs that account for the effect of resettlement locations on integration outcomes. We consider three much debated potential determinants of economic, social, and political integration: the population share of migrants, the rurality of the receiving region, and religious differences between migrants and natives.

To shed light on each of these factors, we exploit quasi-experimental variation in the spatial distribution of refugees, arising from the flight and expulsion of ethnic Germans from Eastern Europe after World War II. About 8 million displaced Germans were resettled in West Germany, where they accounted for 16.5% of the population in 1950. Most importantly for our identification strategy, expellees were distributed very unevenly across West Germany and regardless of their integration prospects.

Using newly digitalized high-quality data at the county level, we embrace a broad concept of integration and investigate local labor market outcomes, marriage market behavior, and voting pattern. Our results show that local conditions play an important role in shaping integration outcomes, and bring a trade-off to light which policymakers in the current "refugee crisis" also face today: On the one hand, refugees should be evenly dispersed across regions as higher population shares of refugees deteriorate economic, social and political integration outcomes. On the other hand, refugees should be sent to more urban areas as these provide significantly better integration prospects. This result cautions against the widely held belief that today's refugees should be mainly sent to rural areas to avoid the formation of ghettos in the cities and to foster rural revival (Bloem 2014, Martínez Juan 2017).

While the literature has so far mainly focused on how the innate characteristics of refugees affect their integration outcomes, this is only part of the story. The decision of policymakers where to re-settle immigrants proves an important driver of integration. Since the resettlement location is a policy variable, while the innate characteristics of refugees are not, the former warrants further analysis. We thus conclude by highlighting the need for future work in the area. In particular, our results are specific to an episode of mass immigration, in which natives and refugees were very similar in many respects, including their mother tongue and cultural background. An important tasks for future work is to assess whether similar findings hold for the resettlement of refugees into religiously and culturally more distant locations.

A Merging of counties

The administrative borders of some West German counties changed between 1939 and 1950. In order to make county borders comparable over time, we first merge counties which, at any time between 1939 and 1950 formed one county. The counties of Hildesheim and Marienburg, for instance, were separate entities in 1939, but were merged to join the new county of Hildesheim-Marienburg in 1946. Consequently, the 1946 and 1950 censuses only contain data on Hildesheim-Marienburg. We thus merge Hildesheim and Marienburg already in the 1939 census. We proceed analogously for the counties of Bremerhaven and Wesermünde; city and rural districts of Bremen; Rhein-Wupper Kreis and Leverkusen; Kreis der Eder, Kreis des Eisenberges and Kreis der Twiste; city and rural districts of Konstanz; Coburg and Rodach bei Coburg; city and rural districts of Dinkelsbühl; city and rural districts of Donauwörth, city and rural districts of Lüneburg.

In addition, there were some smaller border changes, in which municipalities were moved from one county to another. To deal with these border changes, we first compare the 1939 population of each county in its 1950 borders to the 1939 population of the same county in its 1939 borders. Since the majority of administrative borders remained unchanged between 1939 and 1950, the 1939 population figure is usually the same regardless of whether we use 1939 or 1950 borders. Moreover, we do not take any action if the difference between the two population figures is less than 5%. If the difference is larger than 5%, we merge the counties that exchanged municipalities. This applies to the counties of Osterholz, Verden and Bremen; Bergstraße, city and rural districts of Worms; Goslar, Wolfenbüttel and Salzgitter; Mainz, Groß-Gerau and Wiesbaden; Böblingen, Eßlingen and Stuttgart; city and rural districts of Osnabrück; city and rural districts of München; city and rural districts of Kulmbach; Lörrach and Neustadt; Norden and Emden; Braunschweig and Peine.

Finally, we drop counties that have lost or gained more than 5% of its 1939 population to regions outside West Germany, in particular to counties in the Soviet Occupation Zone. These counties include Blankenburg (Rest); Helmstedt; Birkenfeld; Zweibrücken; Saarburg; Trier; Mellrichstadt; Osterode; Lüneburg.

B Data sources

	Table B1: Data sources
Variable	Description and data source
Dependent variables	
Expellee labor force participa- tion rate 1950	The share of economically active persons in the total expellee population in 1950, based on data from Statistische Bundesamt (1955).
Expellee employment rate 1950	Calculated as (100 – Expellee unemployment rate) Expellee labor force participation rate. Data on expellee unemployment rate comes from Pfeil (1958).
Intermarriage rates 1950	Index for intermarriage rates between expellees and non expellees in 1950, taken from Poepelt (1959).
Vote shares of Bayernpartei (and BHE) 1950/54/58	Vote share of Bayernpartei (and BHE) in the Bavarian stat elections of 1950, 1954, and 1958, as published in Bayerische Statistisches Landesamt (1951), Bayerisches Statistische Landesamt (1955), and Bayerisches Statistisches Landesamt (1959).
Main explanatory variables	
Expellee share in 1950	The share of expellees in the 1950 population, based on dat from Statistisches Bundesamt (1952) and Statistisches Bun desamt (1955).
Expellee share in 1946	The share of expellees in the 1946 population, based of Statistisches Bundesamt (1950).
Agricultural employment share in 1939	The share of the workforce in agriculture in 1939, as published in Statistisches Reichsamt (1939)
Religious distance 1950	The Euclidean distance between the religious affiliations of expellees and non-expellees in 1950, based on data from Statistisches Bundesamt (1952). Data on religious affili- ations in 1939, required for calculating the instrument in equation (9), comes from Statistisches Reichsamt (1941).
Control variables	
Population share living in	The 1939 share of population living in cities with at least
cities with at least 10,000 in-	10,000 inhabitants, based on data from Statistisches Reich
habitants 1939	samt (1940).
Rubble per capita 1939	Untreated rubble at the end of the war over the population in 1939, as taken from Deutscher Städtetag (1949).
Damaged dwellings 1945	Share of dwellings built before 1945 damaged in the way based on data from Statistisches Bundesamt (1956).
Distance to inner German	Dummy for whether a county is located within 75 kilometer
border < 75 km $(0/1)$	from the inner-German border.
Majority is Catholic in 1030	Dummy for whether the majority of a county was Cathol

Table B1: Data source

C Additional descriptives





Notes: Scatter plots (a)-(c) show the correlations between our main variables of interest and the labor force participation rate of expellees in 1950. Scatter plots (d)-(f) show the correlations for the marriage index, and scatter plots (g)-(i) show the correlations for the vote share for the *Bayernpartei* as anti-expellee party.

D First stage results

Table D1: First stage results - forced immigration and labor force participation

	(1)	(2)
Dependent variable: Expellee share 1950		
Expellee share 1946	0.879***	0.895***
	(0.017)	(0.027)
Agricultural employment share 1939	-0.001	-0.025***
	(0.008)	(0.009)
Predicted religious distance 1939/1946	-0.002	-0.004
	(0.003)	(0.004)
Population share living in cities with at least 10,000	0.006	-0.007
inhabitants 1939	(0.005)	(0.005)
Rubble per capita 1939	-0.570***	-0.362**
	(0.165)	(0.150)
Distance to inner German border is smaller than 75 km (0/1)	0.007**	0.006*
	(0.003)	(0.003)
Majority is Catholic in 1939 (0/1)	-0.001	-0.001
	(0.002)	(0.002)
Dependent variable: Religious distance 1950		
Expellee share 1946	0.044	0.104
	(0.076)	(0.157)
Agricultural employment share 1939	-0.111**	-0.082*
	(0.044)	(0.043)
Predicted religious distance 1939/1946	0.776***	0.767***
	(0.030)	(0.032)
Population share living in cities with at least 10,000	-0.022	-0.005
inhabitants 1939	(0.015)	(0.022)
Rubble per capita 1939	-1.354*	-1.237*
	(0.694)	(0.729)
Distance to inner German border is smaller than $75 \text{ km} (0/1)$	-0.027*	-0.021
	(0.014)	(0.014)
Majority is Catholic in 1939 (0/1)	0.062***	0.053***
	(0.013)	(0.015)
State dummies	no	yes
Number of observations	526	526

Notes: The table presents the first stage regression results pertaining to columns (6) to (9) of Table 2, columns (3) and (4) of Table 4, columns (5) and (6) of Table 4, columns (6) and (7) of Table 5, and column (2) of Table 6. The first stage regressions use the expellee share in 1946 and the predicted population-weighted religious distance as instrument for the expellee share 1950 and the religious distance 1950, respectively. Column (2) includes dummies for each of the nine West German states. Regressions are weighted with population in 1939. ***, ** and * denote statistical significance at the 1%-, 5%- and 10%-level, respectively. Standard errors clustered at the labor market region level are in parentheses.

References

- Ambrosius, Gerald: 1996, Der Beitrag der Vertriebenen und Flüchtlinge zum Wachstum der westdeutschen Wirtschaft nach dem Zweiten Weltkrieg, Jahrbuch für Wirtschaftsgeschichte / Economic History Yearbook 37(2): 39-72.
- Bauer, Thomas, Sebastian Braun and Michael Kvasnicka: 2013, The Economic Integration of Forced Migrants: Evidence for Post-War Germany, *The Economic Journal* 123(571): 998-1024.
- Bayerisches Statistisches Landesamt: 1950, Die Vertriebenen in Bayern: Ihre berufliche und soziale Eingliederung bis Anfang 1950, Beiträge zur Statistik Bayerns, Heft 151.
- Bayerisches Statistisches Landesamt: 1951, Wahl zum Bayerischen Landtag am 26. November 1950, Beiträge zur Statistik Bayerns, Heft 163.
- Bayerisches Statistisches Landesamt: 1955, Wahl zum Bayerischen Landtag am 28. November 1954, Beiträge zur Statistik Bayerns, Heft 201.
- Bayerisches Statistisches Landesamt: 1959, Wahl zum Bayerischen Landtag am 23. November 1958, Beiträge zur Statistik Bayerns, Heft 211.
- Beaman, Lori A.: 2012, Social Networks and the Dynamics of Labour Market Outcomes: Evidence from Refugees Resettled in the U.S., *Review of Economic Studies* 79(1): 128-161.
- Becker, Sascha O. Becker and Ludger Woessmann: 2009, Was Weber Wrong? A Human Capital Theory of Protestant Economic History, *The Quarterly Journal of Economics* 124(2): 531-596.
- Bloem, Jeffrey: 2014, Refugees in Rural Communities: A Win-Win?, National Agricultural & Rural Development Policy Center Policy Brief 34.
- Brakman, Steven, Harry Garretsen and Marc Schramm: 2004, The Strategic Bombing of German Cities during World War II and its Impact on City Growth, *Journal of Economic Geography* 4(2): 201-218.
- Braun, Sebastian and Michael Kvasnicka: 2014, Immigration and Structural Change: Evidence from Post-war Germany, *The Journal of International Economics* **93(2)**: 253-269.
- Braun, Sebastian and Toman Omar Mahmoud: 2014, The Employment Effects of Immigration: Evidence from the Mass Arrival of German Expellees in Post-war Germany, The Journal of Economic History 74(1): 69-108.
- Braun, Sebastian Till and Henning Weber: 2016, How Do Regional Labor Markets Adjust to Immigration? A Dynamic Analysis for Post-war Germany, *Kiel Working Papers* 2025.
- Brelie-Lewien, Doris von der and Helga Grebing: 1997, 'Flüchtlinge in Niedersachsen', in *Niedersächsische Geschichte*, eds. B.U. Hucker, E. Schubert and B. Weisbrod, 619-634.

- Burchardi, Konrad B. and Tarek A. Hassan: 2013, The Economic Impact of Social Ties: Evidence from German Reunification, *The Quarterly Journal of Economics* 128(3): 1219-1271.
- Connor, Ian: 2007, Refugees and Expellees in Post-War Germany, Manchester University Press.
- Damm, Anna Piil: 2009, Ethnic Enclaves and Immigrant Labor Market Outcomes: Quasi-Experimental Evidence, *Journal of Labor Economics* 27(2): 281-314.
- DER SPIEGEL: 1947, Preußen-Attacke, DER SPIEGEL, 19.04.1947.
- Deutscher Städtetag: 1949, Statistisches Jahrbuch Deutscher Gemeinden 1949, Alfons Bürger Verlag, Schwäbisch Gmünd.
- Douglas, Ray M.: 2012, Orderly and Humane: The Expulsion of the Germans after the Second World War, Yale University Press, New Haven.
- Dustmann, Christian, Kristine Vasiljeva and Anna Piil Damm: 2016, Refugee Migration and Electoral Outcomes, CReAM Discussion Paper Series 16/19.
- Edin, Per-Anders, Peter Fredriksson, and Olof Aslund: 2003, Ethnic Enclaves and the Economic Success of Immigrant–Evidence from a Natural Experiment, *Quarterly Journal of Economics* 118(1): 329-357.
- European Commission: 2015, A European Agenda on Migration, Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions, Brussels.
- Exner, Peter: 1999, 'Integration oder Assimilation? Vertriebeneneingliederung und ländliche Gesellschaft - eine sozialgeschichtliche Mikrostudie am Beispiel westfälischer Landgemeinden', in Geglückte Integration? Spezifika und Vergleichbarkeiten der Vertriebenen-Eingliederung in der SBZ/DDR, eds. Dierk Hoffmann und Michael Schwartz, 57-88.
- Falck, Oliver, Stephan Heblich and Susanne Link: 2012, Forced Migration and the Effects of an Integration Policy in Post-WWII Germany, BE Journal of Economic Analysis & Policy, 12(1): Article 18, 1-27.
- FDZ der Statistischen Ämter des Bundes und der Länder: 2008, Volkszählung 1970, Scientific-Use-File.
- Fernández-Huertas Moraga, Jesús and Hillel Rapoport: 2014, Tradable Refugee-Immigration Quotas, Journal of Public Economics, 115(C): 94-108.
- Fernández-Huertas Moraga, Jesús and Hillel Rapoport: 2015a, Tradable Refugee-Admission Quotas (TRAQs), the Syrian Crisis and the New European Agenda on Migration, IZA Journal of European Labor Studies, 4: 23.
- Fernández-Huertas Moraga, Jesús and Hillel Rapoport: 2015b, Tradable Refugee-admission Quotas and EU Asylum Policy, CESifo Economic Studies, 61(3): 638-672.
- Fiala, Nathan: 2015, The Economic Consequences of Forced Displacement, Journal of Development Studies, 51(10): 1275-1293.

- Grosser, Thomas: 2006, Die Integration der Heimatvertriebenen in Württemberg-Baden (1945-1961), Kohlhammer, Stuttgart.
- Hatton, Timothy J.: 2015, Asylum Policy in the EU: The Case for Deeper Integration, *CESifo Economic Studies*, **61**: 605-637.
- Hatton, Timothy J.: 2016, Refugees, Asylum Seekers and Policy in OECD Countries, American Economic Review (Papers & Proceedings), 106: 441-445.
- Ibáñez, Ana M. and Carlos E. Vélez: 2008, Civil Conflict and Forced Migration: The Micro Determinants and Welfare Losses of Displacement in Colombia, World Development, 36(4): 659-676.
- Institut für Raumfoschung: 1955, Die Kreismappe des Instituts für Raumforschung: Statistische Übersichten, various volumes, Institut für Raumfoschung, Bonn.
- Kondylis, Florence: 2010, Conflict Displacement and Labor Market Outcomes in Post-war Bosnia and Herzegovina, *Journal of Development Economics*, **93(2)**: 235-248.
- Lüttinger, Paul: 1986, Der Mythos der schnellen Immigration. Eine empirische Untersuchung zur Integration der Vertriebenen und Flüchtlinge in der Bundesrepublik Deutschland bis 1971, Zeitschrift für Soziologie, **15(1)**: 20-36.
- Martínez Juan, Ana: 2017, EU Rural Development Policy and the Integration of Migrants, European Parliament Research Service, Brussels.
- Menges, Walter: 1959, 'Wandel und Auflosung der Konfessionszonen', in Die Vertriebenen in Westdeutschland. Ihre Eingliederung und ihr Einfluss auf Gesellschaft, Wirtschaft, Politik und Geistesleben, eds. Eugen Lemberg and Friedrich Edding, Band 3, 1-22.
- MPIDR [Max Planck Institute for Demographic Research] and CGG [Chair for Geodesy and Geoinformatics, University of Rostock]: 2011, MPIDR Population History GIS Collection (partly based on Hubatsch and Klein 1975 ff. and Bundesamt für Kartographie und Geodäsie 2011), Rostock.
- Müller, Werner and Heinz Simon: 1959, 'Aufnahme und Unterbringung', in Die Vertriebenen in Westdeutschland. Ihre Eingliederung und ihr Einfluss auf Gesellschaft, Wirtschaft, Politik und Geistesleben, eds. Eugen Lemberg and Friedrich Edding, Band 1, 300-446.
- Müller, Karl Valentin: 1950, Die Verschwägerung (Konnubium) als soziologischer Maßstab für die Einwurzelung der heimatvertriebenen Bevölkerungsgruppen, Raum und Gesellschaft, 1: 117-133.
- Nellner, Werner: 1959, 'Grundlagen und Hauptergebnisse der Statistik', in Die Vertriebenen in Westdeutschland. Ihre Eingliederung und ihr Einfluss auf Gesellschaft, Wirtschaft, Politik und Geistesleben, eds. Eugen Lemberg and Friedrich Edding, Band 1, 61-144.
- Oltmer, Jochen: 2010, Migration im 19. und 20. Jahrhundert, Oldenbourg, München.
- Pfeil, Elisabeth: 1958, Eingliederungschancen und Eingliederungserfolge: regionalstatistische Analysen der Erwerbstätigkeit, Berufsstellung und Behausung der Vertriebenen, *Mitteilungen aus dem Institut für Raumforschung*, Heft 35.

- Poepelt, Konrad: 1959, Die Verschwägerung der Heimatvertriebenen mit den Westdeutschen, Mitteilungen aus dem Institut für Raumforschung, Verlag Degener & Co, Neustadt an der Aisch.
- Redding, Stephen J. and Daniel M. Sturm: 2008, The Costs of Remoteness: Evidence from German Division and Reunification, *American Economic Review* **98(5)**: 1766-1797.
- Ruiz, Isabel and Carlos Vargas-Silva: 2013, The Economics of Forced Migration, Journal of Development Studies 49(6): 772-784.
- Sarvimäki, Matti, Roope Uusitalo and Markus Jäntti: 2009, Long-Term Effects of Forced Migration, *IZA Discussion Papers 4003*.
- Schulze, Rainer: 2002, 'Wir leben ja nun hier'. Flüchtlinge und Vertriebene in Niedersachsen– Erinnerung und Identität, in Zuwanderung und Integration in Niedersachsen seit dem Zweiten Weltkieg, eds. K.J. Bade and J. Oltmer, 69-100.
- Schulze, Rainer: 2011, 'Forced Migration of German Populations during and after the Second World War: History and Memory', in *The disentanglement of populations. Migration, expulsion and displacement in postwar Europe, 1944-49*, eds. Jessica Reinisch and Elizabeth White, 27-47. Basingstoke: Palgrave Macmillan.
- Schumann, Abel: 2014, Persistence of Population Shocks: Evidence from the Occupation of West Germany after World War II, American Economic Journal: Applied Economics 6(3): 189-205.
- Spiegel-Schmidt, Friedrich: 1959, 'Religiöse Wandlungen und Probleme im evangelischen Bereich', in Die Vertriebenen in Westdeutschland. Ihre Eingliederung und ihr Einfluss auf Gesellschaft, Wirtschaft, Politik und Geistesleben, eds. Eugen Lemberg and Friedrich Edding, 24-91.
- Statistisches Bundesamt: 1946, Die Flüchtlinge in Deutschland. Ergebnisse der Sonderauszählungen aus der Volks- und Berufszählung vom 29. Oktober 1946, *Statistische Berichte*, Arb. Nr. VIII/0/4.
- Statistisches Bundesamt: 1952, Die Bevölkerung der Bundesrepublik Deutschland nach der Zählung vom 13.9.1950, Statistik der Bundesrepublik Deutschland, Band 35.
- Statistisches Bundesamt: 1955a, Einführung in die methodischen und systematischen Grundlagen der Volks- und Berufszählung vom 13. 9. 1950, *Statistik der Bundesrepublik Deutschland*, Band 34.
- Statistisches Bundesamt: 1955b, Die Vertriebenen und Flüchtlinge in der Bundesrepublik Deutschland in den Jahren 1946 bis 1953, *Statistik der Bundesrepublik Deutschland*, Band 114.
- Statistisches Bundesamt: 1956, Gebäude- und Wohnungszählung in der Bundesrepublik Deutschland vom 13. September 1950, Statistik der Bundesrepublik Deutschland, Band 38.
- Statistisches Reichsamt: 1939, Umsatzsteuerstatistik 1935, *Statistik des Deutschen Reichs*, Band 511.

- Statistisches Reichsamt: 1940, Statistisches Jahrbuch für das Deutsche Reich 1939/40, Statistisches Reichsamt, Berlin.
- Statistisches Reichsamt: 1941, Die Bevölkerung des Reichs, der Reichsteile, der größeren und kleineren Verwaltungsbezirke, der Gaue der NSDAP, sowie der Großstädte nach der Religionszugehörigkeit auf Grund der Volkszählung vom 17. Mai 1939., Wirtschaft und Statistik, 21(9).
- Statistisches Reichsamt: 1943, Berufszählung. Die Berufstätigkeit der Bevölkerung in den Reichsteilen. Volks-, Berufs- und Betriebszählung vom 17. Mai 1939, Statistik des Deutschen Reichs, Band 557.
- Stock, J. H. and M. Yogo: 2005, Testing for Weak Instruments in Linear IV Regression, in *Identification and Inference for Econometric Models: Essays in Honor of Thomas Rothenberg*, eds. D. W. K. Andrews and J. H. Stock, 80-108. Cambridge: Cambridge University Press.
- Thielemann, Eiko, Richard Williams, and Christina Boswell: 2010, What System of Burden-Sharing between Member States for the Reception of Asylum Seekers?, Directorate General for Internal Policies, Policy Department C: Citizens' Rights and Constitutional Affairs, Civil Liberties, Justice and Home Affairs, European Parliament, Brussels.
- Weber, Max: 1904/05, Die Protestantische Ethik und der 'Geist' des Kapitalismus, Archiv für Sozialwissenschaft und Sozialpolitik 20: 1-54 and 21: 1-110.
- Ziemer, Gerhard: 1973, Deutscher Exodus. Vertreibung und Eingliederung von 15 Millionen Ostdeutschen, Seewald Verlag, Stuttgart.

Hohenheim Discussion Papers in Business, Economics and Social Sciences

The Faculty of Business, Economics and Social Sciences continues since 2015 the established "FZID Discussion Paper Series" of the "Centre for Research on Innovation and Services (FZID)" under the name "Hohenheim Discussion Papers in Business, Economics and Social Sciences".

Institutes

- 510 Institute of Financial Management
- 520 Institute of Economics
- 530 Institute of Health Care & Public Management
- 540 Institute of Communication Science
- 550 Institute of Law and Social Sciences
- 560 Institute of Economic and Business Education
- 570 Institute of Marketing & Management
- 580 Institute of Interorganisational Management & Performance

Research Areas (since 2017)

INEPA	"Inequality and Economic Policy Analysis"
TKID	"Transformation der Kommunikation – Integration und Desintegration"
NegoTrans	"Negotiation Research - Transformation, Technology, Media and Costs"
INEF	"Innovation, Entrepreneurship and Finance"

Download Hohenheim Discussion Papers in Business, Economics and Social Sciences from our homepage: https://wiso.uni-hohenheim.de/papers

No.	Author	Title	Inst
01-2015	Thomas Beissinger, Philipp Baudy	THE IMPACT OF TEMPORARY AGENCY WORK ON TRADE UNION WAGE SETTING: A Theoretical Analysis	520
02-2015	Fabian Wahl	PARTICIPATIVE POLITICAL INSTITUTIONS AND CITY DEVELOPMENT 800-1800	520
03-2015	Tommaso Proietti, Martyna Marczak, Gianluigi Mazzi	EUROMIND-D: A DENSITY ESTIMATE OF MONTHLY GROSS DOMESTIC PRODUCT FOR THE EURO AREA	520
04-2015	Thomas Beissinger, Nathalie Chusseau, Joël Hellier	OFFSHORING AND LABOUR MARKET REFORMS: MODELLING THE GERMAN EXPERIENCE	520
05-2015	Matthias Mueller, Kristina Bogner, Tobias Buchmann, Muhamed Kudic	SIMULATING KNOWLEDGE DIFFUSION IN FOUR STRUCTURALLY DISTINCT NETWORKS – AN AGENT-BASED SIMULATION MODEL	520
06-2015	Martyna Marczak, Thomas Beissinger	BIDIRECTIONAL RELATIONSHIP BETWEEN INVESTOR SENTIMENT AND EXCESS RETURNS: NEW EVIDENCE FROM THE WAVELET PERSPECTIVE	520
07-2015	Peng Nie, Galit Nimrod, Alfonso Sousa-Poza	INTERNET USE AND SUBJECTIVE WELL-BEING IN CHINA	530

No.	Author	Title	Inst
08-2015	Fabian Wahl	THE LONG SHADOW OF HISTORY ROMAN LEGACY AND ECONOMIC DEVELOPMENT – EVIDENCE FROM THE GERMAN LIMES	520
09-2015	Peng Nie, Alfonso Sousa-Poza	COMMUTE TIME AND SUBJECTIVE WELL-BEING IN URBAN CHINA	530
10-2015	Kristina Bogner	THE EFFECT OF PROJECT FUNDING ON INNOVATIVE PERFORMANCE AN AGENT-BASED SIMULATION MODEL	520
11-2015	Bogang Jun, Tai-Yoo Kim	A NEO-SCHUMPETERIAN PERSPECTIVE ON THE ANALYTICAL MACROECONOMIC FRAMEWORK: THE EXPANDED REPRODUCTION SYSTEM	520
12-2015	Volker Grossmann Aderonke Osikominu Marius Osterfeld	ARE SOCIOCULTURAL FACTORS IMPORTANT FOR STUDYING A SCIENCE UNIVERSITY MAJOR?	520
13-2015	Martyna Marczak Tommaso Proietti Stefano Grassi	A DATA-CLEANING AUGMENTED KALMAN FILTER FOR ROBUST ESTIMATION OF STATE SPACE MODELS	520
14-2015	Carolina Castagnetti Luisa Rosti Marina Töpfer	THE REVERSAL OF THE GENDER PAY GAP AMONG PUBLIC-CONTEST SELECTED YOUNG EMPLOYEES	520
15-2015	Alexander Opitz	DEMOCRATIC PROSPECTS IN IMPERIAL RUSSIA: THE REVOLUTION OF 1905 AND THE POLITICAL STOCK MARKET	520
01-2016	Michael Ahlheim, Jan Neidhardt	NON-TRADING BEHAVIOUR IN CHOICE EXPERIMENTS	520
02-2016	Bogang Jun, Alexander Gerybadze, Tai-Yoo Kim	THE LEGACY OF FRIEDRICH LIST: THE EXPANSIVE REPRODUCTION SYSTEM AND THE KOREAN HISTORY OF INDUSTRIALIZATION	520
03-2016	Peng Nie, Alfonso Sousa-Poza	FOOD INSECURITY AMONG OLDER EUROPEANS: EVIDENCE FROM THE SURVEY OF HEALTH, AGEING, AND RETIREMENT IN EUROPE	530
04-2016	Peter Spahn	POPULATION GROWTH, SAVING, INTEREST RATES AND STAGNATION. DISCUSSING THE EGGERTSSON- MEHROTRA-MODEL	520
05-2016	Vincent Dekker, Kristina Strohmaier, Nicole Bosch	A DATA-DRIVEN PROCEDURE TO DETERMINE THE BUNCHING WINDOW – AN APPLICATION TO THE NETHERLANDS	520
06-2016	Philipp Baudy, Dario Cords	DEREGULATION OF TEMPORARY AGENCY EMPLOYMENT IN A UNIONIZED ECONOMY: DOES THIS REALLY LEAD TO A SUBSTITUTION OF REGULAR EMPLOYMENT?	520

No.	Author	Title	Inst
07-2016	Robin Jessen, Davud Rostam-Afschar, Sebastian Schmitz	HOW IMPORTANT IS PRECAUTIONARY LABOR SUPPLY?	520
08-2016	Peng Nie, Alfonso Sousa-Poza, Jianhong Xue	FUEL FOR LIFE: DOMESTIC COOKING FUELS AND WOMEN'S HEALTH IN RURAL CHINA	530
09-2016	Bogang Jun, Seung Kyu-Yi, Tobias Buchmann, Matthias Müller	THE CO-EVOLUTION OF INNOVATION NETWORKS: COLLABORATION BETWEEN WEST AND EAST GERMANY FROM 1972 TO 2014	520
10-2016	Vladan Ivanovic, Vadim Kufenko, Boris Begovic Nenad Stanisic, Vincent Geloso	CONTINUITY UNDER A DIFFERENT NAME. THE OUTCOME OF PRIVATISATION IN SERBIA	520
11-2016	David E. Bloom Michael Kuhn Klaus Prettner	THE CONTRIBUTION OF FEMALE HEALTH TO ECONOMIC DEVELOPMENT	520
12-2016	Franz X. Hof Klaus Prettner	THE QUEST FOR STATUS AND R&D-BASED GROWTH	520
13-2016	Jung-In Yeon Andreas Pyka Tai-Yoo Kim	STRUCTURAL SHIFT AND INCREASING VARIETY IN KOREA, 1960–2010: EMPIRICAL EVIDENCE OF THE ECONOMIC DEVELOPMENT MODEL BY THE CREATION OF NEW SECTORS	520
14-2016	Benjamin Fuchs	THE EFFECT OF TEENAGE EMPLOYMENT ON CHARACTER SKILLS, EXPECTATIONS AND OCCUPATIONAL CHOICE STRATEGIES	520
15-2016	Seung-Kyu Yi Bogang Jun	HAS THE GERMAN REUNIFICATION STRENGTHENED GERMANY'S NATIONAL INNOVATION SYSTEM? TRIPLE HELIX DYNAMICS OF GERMANY'S INNOVATION SYSTEM	520
16-2016	Gregor Pfeifer Fabian Wahl Martyna Marczak	ILLUMINATING THE WORLD CUP EFFECT: NIGHT LIGHTS EVIDENCE FROM SOUTH AFRICA	520
17-2016	Malte Klein Andreas Sauer	CELEBRATING 30 YEARS OF INNOVATION SYSTEM RESEARCH: WHAT YOU NEED TO KNOW ABOUT INNOVATION SYSTEMS	570
18-2016	Klaus Prettner	THE IMPLICATIONS OF AUTOMATION FOR ECONOMIC GROWTH AND THE LABOR SHARE	520
19-2016	Klaus Prettner Andreas Schaefer	HIGHER EDUCATION AND THE FALL AND RISE OF INEQUALITY	520
20-2016	Vadim Kufenko Klaus Prettner	YOU CAN'T ALWAYS GET WHAT YOU WANT? ESTIMATOR CHOICE AND THE SPEED OF CONVERGENCE	520

No.	Author	Title	Inst
01-2017	Annarita Baldanzi Alberto Bucci Klaus Prettner	CHILDRENS HEALTH, HUMAN CAPITAL ACCUMULATION, AND R&D-BASED ECONOMIC GROWTH	INEPA
02-2017	Julius Tennert Marie Lambert Hans-Peter Burghof	MORAL HAZARD IN VC-FINANCE: MORE EXPENSIVE THAN YOU THOUGHT	INEF
03-2017	Michael Ahlheim Oliver Frör Nguyen Minh Duc Antonia Rehl Ute Siepmann Pham Van Dinh	LABOUR AS A UTILITY MEASURE RECONSIDERED	520
04-2017	Bohdan Kukharskyy Sebastian Seiffert	GUN VIOLENCE IN THE U.S.: CORRELATES AND CAUSES	520
05-2017	Ana Abeliansky Klaus Prettner	AUTOMATION AND DEMOGRAPHIC CHANGE	520
06-2017	Vincent Geloso Vadim Kufenko	INEQUALITY AND GUARD LABOR, OR PROHIBITION AND GUARD LABOR?	INEPA
07-2017	Emanuel Gasteiger Klaus Prettner	ON THE POSSIBILITY OF AUTOMATION-INDUCED STAGNATION	520
08-2017	Klaus Prettner Holger Strulik	THE LOST RACE AGAINST THE MACHINE: AUTOMATION, EDUCATION, AND INEQUALITY IN AN R&D-BASED GROWTH MODEL	INEPA
09-2017	David E. Bloom Simiao Chen Michael Kuhn Mark E. McGovern Les Oxley Klaus Prettner	THE ECONOMIC BURDEN OF CHRONIC DISEASES: ESTIMATES AND PROJECTIONS FOR CHINA, JAPAN, AND SOUTH KOREA	520
10-2017	Sebastian Till Braun Nadja Dwenger	THE LOCAL ENVIRONMENT SHAPES REFUGEE INTEGRATION: EVIDENCE FROM POST-WAR GERMANY	INEPA

FZID Discussion Papers (published 2009-2014)

Competence Centers

IK	Innovation and Knowledge
ICT	Information Systems and Communication Systems
CRFM	Corporate Finance and Risk Management
HCM	Health Care Management
CM	Communication Management
MM	Marketing Management
ECO	Economics

Download FZID Discussion Papers from our homepage: https://wiso.uni-hohenheim.de/archiv_fzid_papers

Nr.	Autor	Titel	CC
01-2009	Julian P. Christ	NEW ECONOMIC GEOGRAPHY RELOADED: Localized Knowledge Spillovers and the Geography of Innovation	IK
02-2009	André P. Slowak	MARKET FIELD STRUCTURE & DYNAMICS IN INDUSTRIAL AUTOMATION	IK
03-2009	Pier Paolo Saviotti, Andreas Pyka	GENERALIZED BARRIERS TO ENTRY AND ECONOMIC DEVELOPMENT	IK
04-2009	Uwe Focht, Andreas Richter and Jörg Schiller	INTERMEDIATION AND MATCHING IN INSURANCE MARKETS	HCM
05-2009	Julian P. Christ, André P. Slowak	WHY BLU-RAY VS. HD-DVD IS NOT VHS VS. BETAMAX: THE CO-EVOLUTION OF STANDARD-SETTING CONSORTIA	IK
06-2009	Gabriel Felbermayr, Mario Larch and Wolfgang Lechthaler	UNEMPLOYMENT IN AN INTERDEPENDENT WORLD	ECO
07-2009	Steffen Otterbach	MISMATCHES BETWEEN ACTUAL AND PREFERRED WORK TIME: Empirical Evidence of Hours Constraints in 21 Countries	HCM
08-2009	Sven Wydra	PRODUCTION AND EMPLOYMENT IMPACTS OF NEW TECHNOLOGIES – ANALYSIS FOR BIOTECHNOLOGY	IK
09-2009	Ralf Richter, Jochen Streb	CATCHING-UP AND FALLING BEHIND KNOWLEDGE SPILLOVER FROM AMERICAN TO GERMAN MACHINE TOOL MAKERS	IK

Nr.	Autor	Titel	CC
10-2010	Rahel Aichele, Gabriel Felbermayr	KYOTO AND THE CARBON CONTENT OF TRADE	ECO
11-2010	David E. Bloom, Alfonso Sousa-Poza	ECONOMIC CONSEQUENCES OF LOW FERTILITY IN EUROPE	HCM
12-2010	Michael Ahlheim, Oliver Frör	DRINKING AND PROTECTING – A MARKET APPROACH TO THE PRESERVATION OF CORK OAK LANDSCAPES	ECO
13-2010	Michael Ahlheim, Oliver Frör, Antonia Heinke, Nguyen Minh Duc, and Pham Van Dinh	LABOUR AS A UTILITY MEASURE IN CONTINGENT VALUATION STUDIES – HOW GOOD IS IT REALLY?	ECO
14-2010	Julian P. Christ	THE GEOGRAPHY AND CO-LOCATION OF EUROPEAN TECHNOLOGY-SPECIFIC CO-INVENTORSHIP NETWORKS	IK
15-2010	Harald Degner	WINDOWS OF TECHNOLOGICAL OPPORTUNITY DO TECHNOLOGICAL BOOMS INFLUENCE THE RELATIONSHIP BETWEEN FIRM SIZE AND INNOVATIVENESS?	IK
16-2010	Tobias A. Jopp	THE WELFARE STATE EVOLVES: GERMAN KNAPPSCHAFTEN, 1854-1923	HCM
17-2010	Stefan Kirn (Ed.)	PROCESS OF CHANGE IN ORGANISATIONS THROUGH eHEALTH	ICT
18-2010	Jörg Schiller	ÖKONOMISCHE ASPEKTE DER ENTLOHNUNG UND REGULIERUNG UNABHÄNGIGER VERSICHERUNGSVERMITTLER	HCM
19-2010	Frauke Lammers, Jörg Schiller	CONTRACT DESIGN AND INSURANCE FRAUD: AN EXPERIMENTAL INVESTIGATION	НСМ
20-2010	Martyna Marczak, Thomas Beissinger	REAL WAGES AND THE BUSINESS CYCLE IN GERMANY	ECO
21-2010	Harald Degner, Jochen Streb	FOREIGN PATENTING IN GERMANY, 1877-1932	IK
22-2010	Heiko Stüber, Thomas Beissinger	DOES DOWNWARD NOMINAL WAGE RIGIDITY DAMPEN WAGE INCREASES?	ECO
23-2010	Mark Spoerer, Jochen Streb	GUNS AND BUTTER – BUT NO MARGARINE: THE IMPACT OF NAZI ECONOMIC POLICIES ON GERMAN FOOD CONSUMPTION, 1933-38	ECO

Nr.	Autor	Titel	CC
24-2011	Dhammika Dharmapala, Nadine Riedel	EARNINGS SHOCKS AND TAX-MOTIVATED INCOME-SHIFTING: EVIDENCE FROM EUROPEAN MULTINATIONALS	ECO
25-2011	Michael Schuele, Stefan Kirn	QUALITATIVES, RÄUMLICHES SCHLIEßEN ZUR KOLLISIONSERKENNUNG UND KOLLISIONSVERMEIDUNG AUTONOMER BDI-AGENTEN	ICT
26-2011	Marcus Müller, Guillaume Stern, Ansger Jacob and Stefan Kirn	VERHALTENSMODELLE FÜR SOFTWAREAGENTEN IM PUBLIC GOODS GAME	ICT
27-2011	Monnet Benoit, Patrick Gbakoua and Alfonso Sousa-Poza	ENGEL CURVES, SPATIAL VARIATION IN PRICES AND DEMAND FOR COMMODITIES IN CÔTE D'IVOIRE	ECO
28-2011	Nadine Riedel, Hannah Schildberg- Hörisch	ASYMMETRIC OBLIGATIONS	ECO
29-2011	Nicole Waidlein	CAUSES OF PERSISTENT PRODUCTIVITY DIFFERENCES IN THE WEST GERMAN STATES IN THE PERIOD FROM 1950 TO 1990	IK
30-2011	Dominik Hartmann, Atilio Arata	MEASURING SOCIAL CAPITAL AND INNOVATION IN POOR AGRICULTURAL COMMUNITIES. THE CASE OF CHÁPARRA - PERU	IK
31-2011	Peter Spahn	DIE WÄHRUNGSKRISENUNION DIE EURO-VERSCHULDUNG DER NATIONALSTAATEN ALS SCHWACHSTELLE DER EWU	ECO
32-2011	Fabian Wahl	DIE ENTWICKLUNG DES LEBENSSTANDARDS IM DRITTEN REICH – EINE GLÜCKSÖKONOMISCHE PERSPEKTIVE	ECO
33-2011	Giorgio Triulzi, Ramon Scholz and Andreas Pyka	R&D AND KNOWLEDGE DYNAMICS IN UNIVERSITY-INDUSTRY RELATIONSHIPS IN BIOTECH AND PHARMACEUTICALS: AN AGENT-BASED MODEL	IK
34-2011	Claus D. Müller- Hengstenberg, Stefan Kirn	ANWENDUNG DES ÖFFENTLICHEN VERGABERECHTS AUF MODERNE IT SOFTWAREENTWICKLUNGSVERFAHREN	ICT
35-2011	Andreas Pyka	AVOIDING EVOLUTIONARY INEFFICIENCIES IN INNOVATION NETWORKS	IK
36-2011	David Bell, Steffen Otterbach and Alfonso Sousa-Poza	WORK HOURS CONSTRAINTS AND HEALTH	HCM
37-2011	Lukas Scheffknecht, Felix Geiger	A BEHAVIORAL MACROECONOMIC MODEL WITH ENDOGENOUS BOOM-BUST CYCLES AND LEVERAGE DYNAMICS	ECO
38-2011	Yin Krogmann, Ulrich Schwalbe	INTER-FIRM R&D NETWORKS IN THE GLOBAL PHARMACEUTICAL BIOTECHNOLOGY INDUSTRY DURING 1985–1998: A CONCEPTUAL AND EMPIRICAL ANALYSIS	IK

Nr.	Autor	Titel	CC
39-2011	Michael Ahlheim, Tobias Börger and Oliver Frör	RESPONDENT INCENTIVES IN CONTINGENT VALUATION: THE ROLE OF RECIPROCITY	ECO
40-2011	Tobias Börger	A DIRECT TEST OF SOCIALLY DESIRABLE RESPONDING IN CONTINGENT VALUATION INTERVIEWS	ECO
41-2011	Ralf Rukwid, Julian P. Christ	QUANTITATIVE CLUSTERIDENTIFIKATION AUF EBENE DER DEUTSCHEN STADT- UND LANDKREISE (1999-2008)	IK

Nr.	Autor	Titel	CC
42-2012	Benjamin Schön, Andreas Pyka	A TAXONOMY OF INNOVATION NETWORKS	IK
43-2012	Dirk Foremny, Nadine Riedel	BUSINESS TAXES AND THE ELECTORAL CYCLE	ECO
44-2012	Gisela Di Meglio, Andreas Pyka and Luis Rubalcaba	VARIETIES OF SERVICE ECONOMIES IN EUROPE	IK
45-2012	Ralf Rukwid, Julian P. Christ	INNOVATIONSPOTENTIALE IN BADEN-WÜRTTEMBERG: PRODUKTIONSCLUSTER IM BEREICH "METALL, ELEKTRO, IKT" UND REGIONALE VERFÜGBARKEIT AKADEMISCHER FACHKRÄFTE IN DEN MINT-FÄCHERN	IK
46-2012	Julian P. Christ, Ralf Rukwid	INNOVATIONSPOTENTIALE IN BADEN-WÜRTTEMBERG: BRANCHENSPEZIFISCHE FORSCHUNGS- UND ENTWICKLUNGSAKTIVITÄT, REGIONALES PATENTAUFKOMMEN UND BESCHÄFTIGUNGSSTRUKTUR	ΙK
47-2012	Oliver Sauter	ASSESSING UNCERTAINTY IN EUROPE AND THE US - IS THERE A COMMON FACTOR?	ECO
48-2012	Dominik Hartmann	SEN MEETS SCHUMPETER. INTRODUCING STRUCTURAL AND DYNAMIC ELEMENTS INTO THE HUMAN CAPABILITY APPROACH	IK
49-2012	Harold Paredes- Frigolett, Andreas Pyka	DISTAL EMBEDDING AS A TECHNOLOGY INNOVATION NETWORK FORMATION STRATEGY	IK
50-2012	Martyna Marczak, Víctor Gómez	CYCLICALITY OF REAL WAGES IN THE USA AND GERMANY: NEW INSIGHTS FROM WAVELET ANALYSIS	ECO
51-2012	André P. Slowak	DIE DURCHSETZUNG VON SCHNITTSTELLEN IN DER STANDARDSETZUNG: FALLBEISPIEL LADESYSTEM ELEKTROMOBILITÄT	IK
52-2012	Fabian Wahl	WHY IT MATTERS WHAT PEOPLE THINK - BELIEFS, LEGAL ORIGINS AND THE DEEP ROOTS OF TRUST	ECO
53-2012	Dominik Hartmann, Micha Kaiser	STATISTISCHER ÜBERBLICK DER TÜRKISCHEN MIGRATION IN BADEN-WÜRTTEMBERG UND DEUTSCHLAND	IK
54-2012	Dominik Hartmann, Andreas Pyka, Seda Aydin, Lena Klauß, Fabian Stahl, Ali Santircioglu, Silvia Oberegelsbacher, Sheida Rashidi, Gaye Onan and Suna Erginkoç	IDENTIFIZIERUNG UND ANALYSE DEUTSCH-TÜRKISCHER INNOVATIONSNETZWERKE. ERSTE ERGEBNISSE DES TGIN- PROJEKTES	ΙK
55-2012	Michael Ahlheim, Tobias Börger and Oliver Frör	THE ECOLOGICAL PRICE OF GETTING RICH IN A GREEN DESERT: A CONTINGENT VALUATION STUDY IN RURAL SOUTHWEST CHINA	ECO

Nr.	Autor	Titel	CC
56-2012	Matthias Strifler Thomas Beissinger	FAIRNESS CONSIDERATIONS IN LABOR UNION WAGE SETTING – A THEORETICAL ANALYSIS	ECO
57-2012	Peter Spahn	INTEGRATION DURCH WÄHRUNGSUNION? DER FALL DER EURO-ZONE	ECO
58-2012	Sibylle H. Lehmann	TAKING FIRMS TO THE STOCK MARKET: IPOS AND THE IMPORTANCE OF LARGE BANKS IN IMPERIAL GERMANY 1896-1913	ECO
59-2012	Sibylle H. Lehmann, Philipp Hauber and Alexander Opitz	POLITICAL RIGHTS, TAXATION, AND FIRM VALUATION – EVIDENCE FROM SAXONY AROUND 1900	ECO
60-2012	Martyna Marczak, Víctor Gómez	SPECTRAN, A SET OF MATLAB PROGRAMS FOR SPECTRAL ANALYSIS	ECO
61-2012	Theresa Lohse, Nadine Riedel	THE IMPACT OF TRANSFER PRICING REGULATIONS ON PROFIT SHIFTING WITHIN EUROPEAN MULTINATIONALS	ECO
Nr.	Autor	Titel	CC
---------	------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------	----------------------------------------------------------------------------------------------------------------------	------
62-2013	Heiko Stüber	REAL WAGE CYCLICALITY OF NEWLY HIRED WORKERS	ECO
63-2013	David E. Bloom, Alfonso Sousa-Poza	AGEING AND PRODUCTIVITY	НСМ
64-2013	Martyna Marczak, Víctor Gómez	MONTHLY US BUSINESS CYCLE INDICATORS: A NEW MULTIVARIATE APPROACH BASED ON A BAND-PASS FILTER	ECO
65-2013	Dominik Hartmann, Andreas Pyka	INNOVATION, ECONOMIC DIVERSIFICATION AND HUMAN DEVELOPMENT	IK
66-2013	Christof Ernst, Katharina Richter and Nadine Riedel	CORPORATE TAXATION AND THE QUALITY OF RESEARCH AND DEVELOPMENT	ECO
67-2013	Michael Ahlheim, Oliver Frör, Jiang Tong, Luo Jing and Sonna Pelz	NONUSE VALUES OF CLIMATE POLICY - AN EMPIRICAL STUDY IN XINJIANG AND BEIJING	ECO
68-2013	Michael Ahlheim, Friedrich Schneider	CONSIDERING HOUSEHOLD SIZE IN CONTINGENT VALUATION STUDIES	ECO
69-2013	Fabio Bertoni, Tereza Tykvová	WHICH FORM OF VENTURE CAPITAL IS MOST SUPPORTIVE OF INNOVATION? EVIDENCE FROM EUROPEAN BIOTECHNOLOGY COMPANIES	CFRM
70-2013	Tobias Buchmann, Andreas Pyka	THE EVOLUTION OF INNOVATION NETWORKS: THE CASE OF A GERMAN AUTOMOTIVE NETWORK	IK
71-2013	B. Vermeulen, A. Pyka, J. A. La Poutré and A. G. de Kok	CAPABILITY-BASED GOVERNANCE PATTERNS OVER THE PRODUCT LIFE-CYCLE	IK
72-2013	Beatriz Fabiola López Ulloa, Valerie Møller and Alfonso Sousa- Poza	HOW DOES SUBJECTIVE WELL-BEING EVOLVE WITH AGE? A LITERATURE REVIEW	НСМ
73-2013	Wencke Gwozdz, Alfonso Sousa-Poza, Lucia A. Reisch, Wolfgang Ahrens, Stefaan De Henauw, Gabriele Eiben, Juan M. Fernández-Alvira, Charalampos Hadjigeorgiou, Eva Kovács, Fabio Lauria, Toomas Veidebaum, Garrath Williams, Karin Bammann	MATERNAL EMPLOYMENT AND CHILDHOOD OBESITY – A EUROPEAN PERSPECTIVE	HCM

Nr.	Autor	Titel	CC
74-2013	Andreas Haas, Annette Hofmann	RISIKEN AUS CLOUD-COMPUTING-SERVICES: FRAGEN DES RISIKOMANAGEMENTS UND ASPEKTE DER VERSICHERBARKEIT	НСМ
75-2013	Yin Krogmann, Nadine Riedel and Ulrich Schwalbe	INTER-FIRM R&D NETWORKS IN PHARMACEUTICAL BIOTECHNOLOGY: WHAT DETERMINES FIRM'S CENTRALITY-BASED PARTNERING CAPABILITY?	ECO, IK
76-2013	Peter Spahn	MACROECONOMIC STABILISATION AND BANK LENDING: A SIMPLE WORKHORSE MODEL	ECO
77-2013	Sheida Rashidi, Andreas Pyka	MIGRATION AND INNOVATION – A SURVEY	IK
78-2013	Benjamin Schön, Andreas Pyka	THE SUCCESS FACTORS OF TECHNOLOGY-SOURCING THROUGH MERGERS & ACQUISITIONS – AN INTUITIVE META- ANALYSIS	IK
79-2013	Irene Prostolupow, Andreas Pyka and Barbara Heller-Schuh	TURKISH-GERMAN INNOVATION NETWORKS IN THE EUROPEAN RESEARCH LANDSCAPE	IK
80-2013	Eva Schlenker, Kai D. Schmid	CAPITAL INCOME SHARES AND INCOME INEQUALITY IN THE EUROPEAN UNION	ECO
81-2013	Michael Ahlheim, Tobias Börger and Oliver Frör	THE INFLUENCE OF ETHNICITY AND CULTURE ON THE VALUATION OF ENVIRONMENTAL IMPROVEMENTS – RESULTS FROM A CVM STUDY IN SOUTHWEST CHINA –	ECO
82-2013	Fabian Wahl	DOES MEDIEVAL TRADE STILL MATTER? HISTORICAL TRADE CENTERS, AGGLOMERATION AND CONTEMPORARY ECONOMIC DEVELOPMENT	ECO
83-2013	Peter Spahn	SUBPRIME AND EURO CRISIS: SHOULD WE BLAME THE ECONOMISTS?	ECO
84-2013	Daniel Guffarth, Michael J. Barber	THE EUROPEAN AEROSPACE R&D COLLABORATION NETWORK	IK
85-2013	Athanasios Saitis	KARTELLBEKÄMPFUNG UND INTERNE KARTELLSTRUKTUREN: EIN NETZWERKTHEORETISCHER ANSATZ	IK

Nr.	Autor	Titel	CC
86-2014	Stefan Kirn, Claus D. Müller-Hengstenberg	INTELLIGENTE (SOFTWARE-)AGENTEN: EINE NEUE HERAUSFORDERUNG FÜR DIE GESELLSCHAFT UND UNSER RECHTSSYSTEM?	ICT
87-2014	Peng Nie, Alfonso Sousa-Poza	MATERNAL EMPLOYMENT AND CHILDHOOD OBESITY IN CHINA: EVIDENCE FROM THE CHINA HEALTH AND NUTRITION SURVEY	HCM
88-2014	Steffen Otterbach, Alfonso Sousa-Poza	JOB INSECURITY, EMPLOYABILITY, AND HEALTH: AN ANALYSIS FOR GERMANY ACROSS GENERATIONS	HCM
89-2014	Carsten Burhop, Sibylle H. Lehmann- Hasemeyer	THE GEOGRAPHY OF STOCK EXCHANGES IN IMPERIAL GERMANY	ECO
90-2014	Martyna Marczak, Tommaso Proietti	OUTLIER DETECTION IN STRUCTURAL TIME SERIES MODELS: THE INDICATOR SATURATION APPROACH	ECO
91-2014	Sophie Urmetzer, Andreas Pyka	VARIETIES OF KNOWLEDGE-BASED BIOECONOMIES	IK
92-2014	Bogang Jun, Joongho Lee	THE TRADEOFF BETWEEN FERTILITY AND EDUCATION: EVIDENCE FROM THE KOREAN DEVELOPMENT PATH	IK
93-2014	Bogang Jun, Tai-Yoo Kim	NON-FINANCIAL HURDLES FOR HUMAN CAPITAL ACCUMULATION: LANDOWNERSHIP IN KOREA UNDER JAPANESE RULE	IK
94-2014	Michael Ahlheim, Oliver Frör, Gerhard Langenberger and Sonna Pelz	CHINESE URBANITES AND THE PRESERVATION OF RARE SPECIES IN REMOTE PARTS OF THE COUNTRY – THE EXAMPLE OF EAGLEWOOD	ECO
95-2014	Harold Paredes- Frigolett, Andreas Pyka, Javier Pereira and Luiz Flávio Autran Monteiro Gomes	RANKING THE PERFORMANCE OF NATIONAL INNOVATION SYSTEMS IN THE IBERIAN PENINSULA AND LATIN AMERICA FROM A NEO-SCHUMPETERIAN ECONOMICS PERSPECTIVE	IK
96-2014	Daniel Guffarth, Michael J. Barber	NETWORK EVOLUTION, SUCCESS, AND REGIONAL DEVELOPMENT IN THE EUROPEAN AEROSPACE INDUSTRY	IK

University of Hohenheim Dean's Office of the Faculty of Business, Economics and Social Sciences Palace Hohenheim 1 B 70593 Stuttgart | Germany Fon +49 (0)711 459 22488 Fax +49 (0)711 459 22785 E-mail wiso@uni-hohenheim.de Web www.wiso.uni-hohenheim.de